

Structural Presumptions for Non-horizontal Mergers in the 2023 Merger Guidelines: A Primer and a Path Forward

Donna, Javier and Pereira, Pedro

University of Florida, Portuguese Competition Authority

20 May 2024

Online at https://mpra.ub.uni-muenchen.de/121615/ MPRA Paper No. 121615, posted 17 Aug 2024 13:53 UTC

Structural Presumptions for Non-Horizontal Mergers in the 2023 Merger Guidelines: A Primer and a Path Forward^{*}

Javier D. Donna^{a,b}

Pedro Pereira^c

May 20, 2024.

Abstract

The 2023 Merger Guidelines (MGs) change the Agencies' narrative regarding non-horizontal mergers. They follow a four-pronged approach: (1) They blend horizontal and non-horizontal mergers. (2) They simplify the narrative about non-horizontal mergers. (3) They consolidate and broaden the theories of harm in non-horizontal mergers. (4) They blend economics and law analysis. In this article, we elaborate on these points. We discuss how the MGs' anticompetitive presumptions apply to non-horizontal mergers, relate them to the economics literature, and provide examples. We finish discussing the economic rationale of the structural presumption involving rivals' exit concerns due to the exercise of market power and propose a path forward.

JEL Codes: K21, K41, L42, L44, L52.

Keywords: Antitrust, 2023 Merger Guidelines, Vertical Mergers, Rivals' Exit, Double Marginalization, Merger Evaluation, Competition Policy.

^aDepartment of Economics, University of Florida, Gainesville, FL, USA. Email: jdonna@ufl.edu. ORCID: https://orcid. org/0000-0002-2383-5444. Corresponding author.

^bRimini Center for Economic Analysis, Rimini, Italy.

^{*}We especially thank Roger D. Blair and David E. M. Sappington for many insightful comments and discussions. We also thank Daron Acemoglu, C. German Bet, Jan Eeckhout, Simon Johnson, Joseph E. Stiglitz, and Gregory F. Veramendi for discussions that have benefitted this article. We thank Pedro A. de Sousa Almeida and Domenica Viteri for excellent research assistance and Sara Benzley, Professor of Legal Research, for her help in acquiring the legal documents that formed the basis for our analysis. We thank the editor, Larry White, for insightful comments and suggestions. Discussions with seminar participants at the Federal Trade Commission have benefited this article. This article has also benefitted from several online discussions—including websites, online publications, social media conversations, authors, and webinars about the 2023 Draft Merger Guidelines and its final document. They include the webinars by the American Enterprise Institute, the CEPR Competition Policy RPN, and the Penn Program on Regulation; the ProMarket Merger Guidelines Symposium; the Sling Economic Analysis and Competition Policy Research; the Thurman Arnold Project at Yale School of Management; and the Utah Project on Antitrust and Consumer Protection; inter alia. We also thank discussions from conference participants and, particularly, the panelists and participants on the special panel about the Draft U.S. Merger Guidelines at the Fifteenth Annual Conference on Antitrust Economics and Competition Policy at Northwestern University, all of which have benefited this article. All opinions, omissions, and errors are our own. The opinions that are expressed in this article reflect only the authors' views and in no way bind the institutions to which they are or were affiliated. The entirety of the production of this article (research, writing, drafting, completion, reviewing, and submission) was done at the time when Javier D. Donna was affiliated with the University of Florida and Pedro Pereira was affiliated with the Portuguese Competition Authority. The affiliations in the article only reflect the affiliations at the time of the publication, not production. Declaration of Conflicting Interests: The authors declare no potential conflicts of interest with respect to the research, authorship, and/or publication of this article. The authors do not work for nor advise any private party with financial interests in the Merger Guidelines. Funding: The authors received no financial support for the research, authorship, and/or publication of this article. First public draft of the article: February 5, 2024.

^cAutoridade da Concorrência, Lisbon, Portugal. Email: pedro.br.pereira@gmail.com. ORCID: https://orcid.org/0000-0002-9036-0354.

1 Introduction

The 2023 Merger Guidelines (MGs) change the Agencies' narrative with regard to non-horizontal mergers.¹ They follow a four-pronged approach: First, they blend horizontal and non-horizontal mergers.² Second, they simplify the narrative about non-horizontal mergers, emphasizing that the Agencies do not need to prove the anticompetitive results with certainty.³ Third, they consolidate and broaden the theories of harm in non-horizontal mergers; they leverage on the latest advances from the economics literature. Finally, they blend economics and law analysis.

In this article, we elaborate on these points. We discuss how the MGs' anticompetitive presumptions apply to non-horizontal mergers, relate them to the economics literature, and provide examples. We finish by discussing in more depth the economic rationale of the structural presumption against non-horizontal mergers that cause harm due to rivals' exit and propose a path forward.⁴

The words "may be substantially to lessen competition, or to tend to create a monopoly"⁵ are the common denominator of the novel four-path narrative. These words, from Section 7 of the Clayton Act, are the words that were used by Congress in 1914.⁶ According to the Supreme Court, they "indicate that its concern was with probabilities, not certainties."⁷ Congress' probability approach "creates a relatively expansive definition of antitrust liability: To show that a merger is unlawful, a plaintiff need only prove that its effect "may be substantially to lessen competition."⁸ The antitrust liability is expanded when bundled with the words "in

³"No person engaged in commerce or in any activity affecting commerce shall acquire, directly or indirectly, the whole or any part of the stock or other share capital and no person subject to the jurisdiction of the Federal Trade Commission shall acquire the whole or any part of the assets of another person engaged also in commerce or in any activity affecting commerce, where in any line of commerce or in any activity affecting commerce in any section of the country, the effect of such acquisition may be substantially to lessen competition, or to tend to create a monopoly." 15 U.S.C. §18.

⁴We define the term *structural presumption* as a proposition that the results of a merger (horizontal or non-horizontal) are likely to be anticompetitive and thus the merger should be deemed illegal unless the presumption is rebutted. "The structural presumption is so named because the likely anticompetitive effects of such mergers are in some sense presumed to follow from the *change in market structure* involved in such consolidations." (Emphasis added; Sullivan 2016, p. 405, referring to horizontal mergers. Also related to horizontal mergers, see Kaplow and Shapiro 2007, pp. 17-83.) For discussions, see Baker and Shapiro (2008), Salop (2015), Sullivan (2016), and the references therein. We use the terms *structural* and *anticompetitive* presumptions interchangeably.

⁵Emphasis added; 15 U.S.C. §18.

⁶The original 1914 text reads: "SEC. 7. That no corporation engaged in commerce shall acquire directly or indirectly, the whole or any part of the stock or other share capital of another corporation engaged also in commerce, where the effect of such acquisition may be to substantially lessen competition between the corporation whose stock is so acquired and the corporation making the acquisition, or to restrain such commerce in any section or community, or tend to create a monopoly of any line of commerce." (Emphasis added; Sixty-Third Congress 1914, Clayton Act, Sess, II, Ch. 323, §7, ¶1, pp. 731-732.) See Collins (2016) for a discussion.

⁷Brown Shoe Co. v. U.S., 370 U.S. 294 (1962), p. 323.

¹U.S. Department of Justice and the Federal Trade Commission (2023) (henceforth, MGs). We use the term Agencies to refer collectively to all government agencies and regulators that might review a merger.

²Non-horizontal mergers include vertical, complementary, product-extension, conglomerate, and platform mergers, inter alia.

⁸California v. American Stores Co., 495 U.S. 271 (1990), p. 284.

any line of commerce or in *any* activity affecting commerce in *any* section of the country."⁹ In economic words, in any market.

For example, in a recent decision that blocked the JetBlue-Spirit (horizontal) merger, the District Court stated: "Section 7 does not require proof that a merger or other acquisition has caused higher prices in the affected market. All that is necessary is that the merger create an appreciable danger of such consequences in the future."¹⁰ The same principles apply to all mergers because *lessening competition is horizontal in nature*—regardless of the merger label, harm theory, market, and industry. The structural presumptions in the MGs reflect the expansive definition of antitrust liability from the probability approach in the Clayton Act. They may also reflect a response to the decline in antitrust enforcement in the United States since the 1960s.¹¹

Merger label removal. The first point is that the MGs remove the horizontal and nonhorizontal merger labels. The label removal is consistent with the wording in the law that the Agencies enforce. The blending helps the Agencies develop new lines to challenge anticompetitive mergers. The MGs are broader than the 2010 Horizontal Merger Guidelines.¹² They incorporate and update the 2020 Vertical Merger Guidelines.¹³

This approach is helpful for the Agencies for three reasons. First, whereas the distinction between horizontal and non-horizontal mergers is clear in the textbooks and theoretical models, it is less obvious in the real-world cases that are challenged by the Agencies. The difficulty is widespread in the modern economy and digital markets, where dominant firms may participate in several markets, product lines, platforms, and ecosystems.

Second, vertical and complementary-product mergers have an intrinsic efficiency, the elimination of the double marginalization (EDM).¹⁴ Starting with such an efficiency presumption

⁹Emphasis added; 15 U.S.C. §18.

¹⁰United States and Plaintiff States v. JetBlue Airways Corporation and Spirit Airlines, Inc. (2024), p. 63 (quoting Saint Alphonsus Med. Ctr.-Nampa Inc., 778 F.3d, 788 quoting Hospital Corp. of Am. v. F.T.C., 807 F.2d 1381, 1389, 7th Cir. 1986).

¹¹See Lancieri, Posner and Zingales (2022). Kwoka (2015) presents a thorough analysis of merger retrospectives and finds unambiguous evidence that most of the studied mergers resulted in higher prices and competitive harm. For additional discussions, see Shapiro (2018, 2019), and the references therein.

¹²U.S. Department of Justice and the Federal Trade Commission (2010) (henceforth, HMGs).

¹³U.S. Department of Justice and the Federal Trade Commission (2020) (henceforth, VMGs).

¹⁴For complementary mergers, the result dates back to Cournot (1838). Intuitively, consider the case where one firm, the acquirer, merges with another firm, the acquiree. Before the merger, the acquirer and acquiree do not account for the effect of their markup on the other firm, thus reducing the demand for the other firm's complementary product. After the merger, the merged firm internalizes the reduction in the demand for the acquirer's products caused by the increase in the price of the acquiree's products, and vice versa. The resulting increase in efficiency is similar to the one resulting from the elimination of double marginalization in a vertical merger. See Alderman, Blair and Donna (2024) for a recent discussion.

hinders the Agencies' efforts to investigate the specific details of the merger.¹⁵ Rather than focusing on the label and starting with a procompetitive efficiency—which may not necessarily be merger-specific nor benefit consumers—, the MGs focus on harm and the likelihood that the merger might substantially lessen competition. This point is useful for the Agencies in mergers that feature both horizontal and non-horizontal elements.

The third is perhaps the most fundamental point: "All so-called vertical merger cases should be handled through the application of horizontal merger standards."¹⁶ The basic principle that governs the Agencies' concern in non-horizontal mergers is their horizontal harm. Thus, merging horizontal and non-horizontal guidelines is sensible from a policy standpoint to enhance antitrust enforcement.¹⁷

Simplified narrative and risk assessment. The second point is that the MGs simplify the narrative with regard to non-horizontal mergers. Non-horizontal mergers are notoriously complex. The Agencies have faced difficulties in the courts defending the theories of anticompetitive harm in non-horizontal mergers. Examples include proving raising rivals' costs, foreclosure, and bargaining leverage effects.¹⁸ The simplified approach allows the Agencies to deal with the uncertain nature of the merger effects in non-horizontal markets. The Agencies specified this approach at the beginning of the MGs:

"To show that a merger is unlawful, a plaintiff need only prove that its effect 'may be substantially to lessen competition'" or to tend to create a monopoly.¹⁹ Accordingly, the Agencies do not attempt to predict the future or calculate precise effects of a merger with certainty. Rather the Agencies examine the totality of the evidence available to assess the risk the merger presents. (Emphasis in original; bold added; MGs, pp. 1-2; footnote included.)

It is a subtle but fundamental change relative to the HMGs and the VMGs. It has consequential implications for economic modeling. The role of the models is not to predict a number

¹⁶Bork (1978, pp. 244-245).

¹⁵To be sure, we are not saying that the EDM is not present in vertical or complementary mergers. Instead, we are emphasizing the point made by the Agencies. "The Agencies examine whether elimination of double marginalization satisfies the approach to evaluating procompetitive efficiencies in Section 3.3, including examining: (a) whether the merged firm will be more vertically integrated as a result of the merger, for example, because it increases the extent to which it uses internal production of an input when producing output for the relevant market; (b) whether contracts short of a merger have eliminated or could eliminate double marginalization such that it would not be merger-specific, and (c) whether the merged firm has the incentive to reduce price in the relevant market given that such a reduction would reduce sales by the merged firm's rivals in the relevant market, which would in turn lead to reduced revenue and margin on sales of the related product to the dependent rivals." (MGs, footnote 31.) See Kwoka and Slade (2019) for a discussion.

¹⁷Why is the focus of this article on non-horizontal mergers? Whereas the blending (of horizontal and non-horizontal mergers) is useful for antitrust enforcement policy, the distinction is helpful from a modeling standpoint in economic theory. It allows the development and isolatation of specific anticompetitive effects and theories of harm, as we show in Section 5.

¹⁸See Salop and Scheffman (1983), Rogerson (2020) Shapiro (2021), Donna et al. (2024), Alderman, Blair and Donna (2024), and the references therein.

¹⁹Footnote 6, MGs: "quoting *California v. Am. Stores Co.*, 495 U.S. 271, 284 (1990) (quoting 15 U.S.C. §18 with emphasis) (citing *Brown Shoe*, 370 U.S. at 323)."

with certainty—a specific price increase or quantity decrease—but to perform a risk assessment. The focus on risk assessment allows the Agencies to evaluate all of the evidence that they have. For example, the evidence could be a theoretical model that is combined with an empirical analysis and qualitative evidence. Simultaneously, the Agencies recognize the difficulty of performing detailed merger simulations in some instances. This avenue could help assess the harm from future events that are difficult to predict with certainty—such as innovation and rivals' entry and exit decisions.

Expansion of harm theories. The third point relates to expanding the harm theories in non-horizontal mergers. A significant innovation in the (withdrawn) 2020 VMGs was to replace the obsolete portions of the 1984 (Non-Horizontal) Merger Guidelines. The VMGs included theories of anticompetitive harm in vertical mergers. The VMGs recognized that a vertical merger might harm competition due to firm conduct—such as foreclosure, raising rivals' costs, and rivals' exit. Notably, these theories of harm were absent in the 1984 Guidelines.

The MGs preserve and broaden this path. They take advantage of the theoretical and empirical advances in the economics literature.²⁰ As we will discuss in Sections 2 and 5, the MGs expand the breadth and depth of the harm theories in non-horizontal markets. They now include the VMGs' traditional harm theories and modern ones, such as entrenchment, dynamics, serial acquisitions, and platforms.²¹ Notably, the extent to which rivals' exit may produce horizontal harm in non-horizontal mergers is prominently featured.²²

Blending of economic and law analysis. The final point is that the MGs combine economic and law technical analyses. Nevertheless, the MGs are centered on antitrust enforcement policy: The MGs highlight how the Agencies enforce current law. They present the prevailing statutory and case law and describe how the Agencies use the analytical, economic, and evidentiary tools. The Agencies use modern economic analysis to articulate harm theories and to discuss how they enforce antitrust law.

However, the MGs do not include references to the economics literature. This approach reflects the Agencies' position: Merger enforcement policy belongs in the policy domain, not in the technical domain.

To complement the MGs' contributions to antitrust enforcement policy, we provide: (1) An introduction to the economic literature that underlies the MGs, with a focus on non-horizontal mergers; and (2) an economic analysis of the structural presumption against non-horizontal mergers that could lead rivals' exit due to market power. By covering the latter aspect in more detail, we illustrate how economic analysis highlights critical issues for future developments in antitrust policy.

On the one hand, the theories of harm for horizontal mergers focus on the potential of

²⁰See Rey and Tirole (2007), Lee, Whinston and Yurukoglu (2021), and Asker and Nocke (2021) for surveys. See Salop (2018) for a discussion.

 $^{^{21}}$ Guidelines 6, 7, 8, and 9.

²²See Appendix A.

the transaction to increase the *ability and incentive* of the merging firms to exercise market power. There are many tools and extensive experience to evaluate this possibility and quantify its impact on prices and social welfare. On the other hand, the theories of harm for vertical mergers have focused on the potential of the transaction to increase the *ability and incentive* of the firms to foreclose rivals or raise rivals' costs, which can be subsumed in exclusionary behavior.²³ This circumstance creates a schism between the review of horizontal and vertical mergers.

Such dichotomy is unnecessary to evaluate one specific type of non-horizontal mergers: Those that may be particularly harmful due to exit concerns.²⁴ The mentioned schism complicates the harm narrative in these mergers. First, vertical mergers may increase the ability and incentive of the merging firms to exercise market power through the vertical-diversion effect, as we show in Section 5. Foreclosing rivals can be seen as a limiting case of the exercise of market power. Even if a vertically integrated firm is unable or unwilling to foreclose rivals, it may still be able and willing to cause harm through the exercise of market power. However, establishing the ability and incentive to foreclose rivals is harder than establishing the ability and incentive to exercise market power. This is one of the difficulties that the Agencies have faced in non-horizontal mergers, as was discussed above.

Second, the exercise of market power and the efficiencies that are enabled by vertical mergers may cause—perhaps unintentionally—rivals to exit the market.²⁵ This possibility can considerably increase the harmful nature of a vertical merger. A vertical merger that causes rivals' exit might reduce welfare even if it is a welfare-enhancing merger absent exit. However, an evaluation of a vertical merger that is based only on the ability and incentive for exclusionary behavior might miss the probability of rivals' exit that is caused by the exercise of market power or efficiencies.

To be sure, we do not advocate against the exclusionary theories of harm. On the contrary, these theories of harm are and should remain central in the Agencies' toolkit. Our theory of harm does not preclude foreclosure nor raising rivals' costs effects. Both theories are complementary in our view. Because the presence of exclusionary or predatory conduct might raise rivals' costs, it further increases the probability that a rival might exit.

Thus, both theories of harm ought to be considered by the Agencies as appropriate in the industries and cases under consideration. We propose that, for the mentioned mergers, the Agencies shift the emphasis of the evaluation to an easier-to-prove theory of harm with a focus on the possibility and consequences of rivals' exit. This harm theory makes it easier for the Agencies to establish harm and explain it to the courts. It provides a unified approach to evaluating horizontal and vertical mergers and blends the four points above.

²³For concreteness, we focus our discussion here on vertical mergers, noting that the discussion applies to non-horizontal mergers with the appropriate differences.

²⁴Rivals' exit is not required to establish harm in non-horizontal mergers. Nevertheless, if exit occurs, the possibility of harm increases substantially. See Donna and Pereira (2023a) for examples and a discussion.

²⁵That a vertical merger may reduce competition *even absent any specific anticompetitive intent* is missing in the MGs.

The rest of this article is organized as follows. Section 2 discusses how the MGs' structural presumptions fit non-horizontal mergers, relates them to the economic literature, and presents examples. Then, the article focuses on the structural presumptions that involve rivals' exit concerns. Section 3 discusses these concerns in the MGs and presents the terminology. Section 4 presents the model. Section 5 discusses the theories of benefit and harm. Section 6 concludes.

2 Structural Presumptions for Non-Horizontal Mergers

We now discuss how the MGs' structural presumptions fit non-horizontal mergers. As we mentioned in the Introduction, the structural presumptions apply to all mergers: Horizontal and non-horizontal.

The economics literature has identified several paths through which non-horizontal mergers may harm competition.²⁶ The MGs incorporate recent developments from this literature and open doors for future expansion.²⁷ We elaborate below.

The MGs are structured as follows: Sections 1 and 2 present and discuss the application of 11 Guidelines. The Guidelines identify mergers that might harm competition. The purpose of the Guidelines is "to assist the Agencies in assessing whether a merger presents sufficient risk to warrant an enforcement action. [...] Guidelines 1-6 describe distinct frameworks the Agencies use to identify that a merger raises prima facie concerns, and Guidelines 7-11 explain how to apply those frameworks in several specific settings."²⁸ Crucially, "[t]hese Guidelines are not mutually exclusive" and "they are not exhaustive."²⁹

The Guidelines focus on situations when "companies propose a merger that raises concerns,"³⁰—not on the positive aspects of the merger. The Guidelines can be rebutted, as explained in Section 3 about rebuttal evidence. According to the MGs, it is the firms' burden to rebut the anticompetitive presumptions. The analytical, economic, and evidentiary tools are presented in Section 4.

In the remainder of this section, we discuss each of the Guidelines as follows: First, we introduce and relate the Guideline to the underlying economics literature. Second, we discuss how the Guidelines may incorporate a structural presumption in non-horizontal mergers and how the Agencies may apply it. Finally, we present examples of its application.

The MGs introduce modern economics theories using current law. Nevertheless, the MGs do not include references to the economics literature—just references to statutory and case law.

²⁶See Salop (1981), Salop and Scheffman (1983), Salinger (1988), Bernheim and Whinston (1998), Segal and Whinston (2000), Motta (2004, Chapter 6), Rey and Tirole (2007), Scheffman and Higgins (2014), and the references therein. See also the vertical integration studies in Chipty (2001), Hastings (2004), Hortaçsu and Syverson (2007), and Houde (2012); and the more recent discussions by Salop (2018), Beck and Scott Morton (2021), and Moresi and Salop (2021). For a recent survey of the academic literature on antitrust economics, see Asker and Nocke (2021).

 $^{^{27}}$ Inter alia, Crawford et al. (2018), Luco and Marshall (2020), Donna and Pereira (2023 a).

 $^{^{28}}MGs$, p. 2.

²⁹Ibid., p. 2 and p. 29, respectively.

 $^{^{30}\}mathrm{Ibid.},$ p. 2.

We fill this gap by providing an introduction to the economics literature that underlies the MGs centered on non-horizontal mergers. Our focus is on a short introductory piece to this literature rather than a comprehensive review of a few prominent papers. Thorough analyses and comprehensive literature reviews can be found in the references. Similarly, a discussion on horizontal mergers, rebuttal evidence, and the analytical, economic, and evidentiary tools is beyond the scope of this article.

Guideline 1 states that "Mergers Raise a Presumption of Illegality When They Significantly Increase Concentration in a Highly Concentrated Market."^{31,32} Guideline 1 targets mergers that are particularly harmful: For example, a 3-to-2 horizontal merger or a vertical merger that poses a similar risk because it may cause a rival's exit in such a concentrated market.³³ Guideline 1 alone might raise anticompetitive concerns in non-horizontal mergers: *E.g.*, "if the merger threatens to cause the exit of a current market participant."³⁴

Therefore, Guideline 1 introduces a structural presumption for non-horizontal mergers in highly concentrated markets.³⁵ But Guideline 1 may also be combined with any (or multiple) of the other guidelines. For example, Guideline 1 may be combined with Guideline 5 in the case of a vertical merger in a highly concentrated market where there is a concern that the merging companies may "limit access to products or services that its [horizontal] rivals use to compete."³⁶ We discuss the other guidelines in turn, noting that they may be combined among themselves due to their non-exclusive nature.

Examples of applications of the language of Guideline 1 include: United States v. Philadelphia Nat'l Bank, 374 U.S. 321 (1963); United States v. Alcoa, 377 U.S. 271 (1964); Federal Trade Commission v. Hackensack Meridian Health Inc, No. 21-2603 (3d Cir. 2022); and United States v. AT&T, Inc., 916 F.3d 1029 (2019).

Guideline 2 introduces an anticompetitive presumption for mergers that "Eliminate Substantial Competition Between Firms."^{37,38} In the case of non-horizontal mergers, where the merging firms may not be directly competing, this situation may arise through the effect on the merging firms' rivals. An example would be a vertical or product-extension merger, where the merging firms may take "competitive actions to attract customers [...] at the expense of

 $^{^{31}}$ Ibid., p. 5.

³²There has been a growing concern in recent years about rising prices, increases in industry concentration, and rising markups. For discussions, see Gutiérrez and Philippon (2017), Berry, Gaynor and Scott Morton (2019), De Loecker, Eeckhout and Unger (2020), Autor et al. (2020), Eeckhout (2022)), and the references therein. Also related to Guideline 1, see the investigations in Ashenfelter, Hosken and Weinberg (2013), Kwoka (2016), Fan and Yang (2020), Cooper et al. (2019), Wollmann (2019), Benkard, Yurukoglu and Zhang (2021), Nocke and Whinston (2022), and the references therein.

 $^{^{33}}$ Donna and Pereira (2023*a*) present a theoretical framework and examples for these vertical mergers. 34 MGs, footnote 11.

 $^{^{35}}$ "Markets with an HHI greater than 1,800 are highly concentrated, and a change of more than 100 points is a significant increase." (Ibid., p. 5.)

 $^{^{36}\}mathrm{Ibid.},$ p. 3.

 $^{^{37}}$ Ibid, p. 6

³⁸For references in the economic literature, see Wollmann (2019), Wollmann (2020), and Bhattacharya, Illanes and Stillerman (2023).

its rivals."³⁹

Examples include: Brown Shoe Co. v. U.S., 370 U.S. 294 (1962); United States v. First Nat'l Bank of Lexington, 376 U.S. 665 (1964); and FTC v. Procter & Gamble Co., 386 U.S. 568 (1967).⁴⁰

Guideline 3 focuses on a structural presumption for mergers that "Increase the Risk of Coordination."^{41,42} For instance, this situation may occur when a vertical merger "eliminates a maverick or significantly changes its incentives increas[ing] the susceptibility to coordination."⁴³

Examples of mergers that may facilitate the risk of coordination are *Brooke Group Ltd. v.* Brown & Williamson Tobacco Corp., 509 U.S. 209 (1993), and joint ventures, as is discussed in Guideline 11.

Guideline 4 incorporates a structural presumption for mergers that "Eliminate a Potential Entrant in a Concentrated Market."^{44,45} The argument is that such a "merger can also eliminate current competitive pressure exerted on other market participants by the mere perception that one of the firms might enter",⁴⁶ which is supported by economic theory.

Guideline 4 is relevant for mergers that involve non-horizontal markets. An example triggering this presumption in non-horizontal mergers may arise when a dominant firm acquires several small, innovative competitors in those markets potentially offending Section 7 of the Clayton Act, but also a potential unlawful monopolization of the market in violation of Section 2 of the Sherman Act.⁴⁷

Some examples include: United States v. Penn-Olin Chemical Co., 378 U.S. 158 (1964); Ford Motor Co. v. United States, 405 U.S. 562 (1972); and United States v. Marine Bancorporation, Inc., 418 U.S. 602 (1974); inter alia.

Recent examples include: Federal Trade Commission v. Facebook (2021), the first amended complaint against Facebook, where the FTC argued that "Facebook's strategy to prevent innovative entrants from gaining scale and benefiting from network effects has consisted of acquiring innovators and—where possible—transforming their products into integral parts of the company's competitive 'moat', "⁴⁸ and Federal Trade Commission v. Meta Platforms et al. (2022),

 $^{^{39}\}mathrm{MGs},$ p. 7.

 $^{^{40}}$ Ibid, p. 7. See Donna and Pereira (2023*a*) for additional examples.

 $^{^{41}}MGs$, p. 8.

⁴²Studies in the economic literature include, *e.g.*, Nocke and White (2007), Miller and Weinberg (2017), Grieco, Pinkse and Slade (2018), Federico, Langus and Valletti (2018), Salop (2018), Miller, Sheu and Weinberg (2021), Igami and Sugaya (2022), and Aryal, Ciliberto and Leyden (2022).

⁴³MGs, p. 9. For a discussion of coordinated effects in vertical mergers, see the VMGs (Section 5) and Salop (2020, Section 5).

 $^{^{44}}MGs$, p. 10.

⁴⁵There is a large economic literature investigating potential competition, entry deterrence, limit pricing, and killer acquisitions. See Kwoka and Shumilkina (2010), Li et al. (2015), Cunningham, Ederer and Ma (2021), Caradonna, Miller and Sheu (2023), Ciliberto, Murry and Tamer (2021), Fan and Yang (2022), and the references therein.

⁴⁶MGs, p. 11.

 $^{^{47}15}$ U.S.C. \$18 and 15 U.S.C. \$2, respectively

⁴⁸Federal Trade Commission v. Facebook (2021), ¶66. See also Scott Morton and Dinielli (2022).

where the complaint challenged Meta acquisition of Within.⁴⁹

Guideline 5 states that "Mergers Can Violate the Law When They Create a Firm that May Limit Access to Products or Services That Its Rivals Use to Compete."⁵⁰ The structural presumption in Guideline 5 incorporates well-known theories of harm from mergers in non-horizontal markets, including: Raising rivals' costs;⁵¹ foreclosure;⁵² exclusion;⁵³ quality degradation;⁵⁴ and rivals' exit.⁵⁵

In particular, the structural presumption with regard to rivals' exit is more prominently featured relatively to the VMGs: "Rivals or potential rivals may be excluded from the relevant market if limiting their access to the related product *could lead them to exit the market* or could deter them from entering."⁵⁶ This structural presumption is similar to the one suggested by Donna and Pereira (2023*a*), Donna and Pereira (2023*b*), and Donna and Pereira (2023*c*). The Agencies acknowledge the difficulty of predicting these events with certainty and note that "the Agencies focus on the overall risk that the merged firm will do so, and do not necessarily identify which precise actions the merged firm would take to lessen competition."⁵⁷

The Agencies inquire about the incentive and ability of the merging firms to foreclose rivals by assessing four factors: (i) "[T]he availability of substitutes for the related product";⁵⁸ (ii) "how important the related product is for the dependent firms";⁵⁹ (iii) "the importance of the dependent firms for competition in the relevant market";⁶⁰ and (iv) "the competition between the merged firm and the dependent firms."⁶¹

Notably, footnote 30 introduces an anticompetitive presumption with regard to the structure of the related market in non-horizontal mergers: "The Agencies will generally infer, in the absence of countervailing evidence, that the merging firm has or is approaching monopoly power in the related product if it has a share greater than 50% of the related product market. A

⁴⁹The Court accepted the potential competition argument but imposed an "excessive evidentiary burden," according to Salop (2023). See also the discussion by Singer (2023), who was the FTC's economic expert.

⁵²Allen (1971), Hart and Tirole (1990), Ordover, Saloner and Salop (1990), Hortaçsu and Syverson (2007), Asker (2016), inter alia. Rey and Tirole (2007) provide a thorough analysis and review of the literature on foreclosure. For a recent case, see Illumina, Inc. v. FTC, No. 23-60167, (5th Cir. Dec. 15, 2023).

⁵³See Marvel (1982), Rasmusen, Ramseyer and Wiley (1991), the subsequent comment by Segal and Whinston (2000), the extension by Fumagalli and Motta (2006), Nocke and Rey (2018), and the references therein. ⁵⁴See Choi and Kim (2010) and Allain, Chambolle and Rey (2016).

 $^{^{50}}MGs$, p. 13.

⁵¹Salop and Scheffman (1983) and Scheffman and Higgins (2014).

⁵⁵Donna and Pereira (2023*a*) present a model and examples. Alderman, Blair and Donna (2024) discuss the recent Microsoft-Activision acquisition and present a theoretical model for the case of complement mergers. As discussed by Donna and Pereira (2023*a*), the idea that vertical integration might induce rivals' exit is not new. It dates back, at least, to Brown Shoe (1962). Mainstream antitrust and economic scholars have long recognized that non-horizontal schemes may harm horizontal rivals (*e.g.*, see Areeda and Hovenkamp 2015-2021, $\P1000-\P1041$, $\P1600-\P1619$, $\P1800-\P1823$; Bork 1978, pp. 152-164 and 237-252; Motta 2004, Chapter 6; and Rey and Tirole 2007). See Donna and Pereira (2023*a*) for details.

⁵⁶Emphasis added; MGs, p. 14.

⁵⁷Ibid., p. 14.

⁵⁸Ibid., p. 14.

⁵⁹Ibid., p. 14.

⁶⁰Ibid., p. 14.

⁶¹Ibid., p. 14.

merger involving a related product with share of less than 50% may still substantially lessen competition, particularly when that related product is important to its trading partners."⁶²

Guideline 5 is comprehensive in that it includes "products rivals currently or may in the future use as inputs,"⁶³ complements,⁶⁴ "access to competitively sensitive information,"⁶⁵ and bargaining situations "[where product availability] enables rivals to obtain better terms from other providers in negotiations,"⁶⁶ in cases "whether or not they involve a traditional vertical relationship."⁶⁷

Examples of application of Guideline 5 include: Brown Shoe Co. v. U.S., 370 U.S. 294 (1962); FTC v. Procter & Gamble Co., 386 U.S. 568 (1967); and Fruehauf Corp. v. FTC, 603 F.2d 345 (2d Cir. 1979); inter alia. A recent example is provided by Illumina v. FTC, No. 23-60167 (5th Cir. 2023).

The MGs highlight that Guideline 5 may be combined with Guidelines 2, 3, 4, and 8 (Series of Multiple Acquisitions).⁶⁸

Guidelines 6, 7, and 8 incorporate dynamics into merger evaluation. Rather than assessing a one-shot industry picture, the Agencies open the door to investigate competition as a dynamic phenomenon, inquiring about where the industry is coming and where it is going.⁶⁹

Guideline 6 introduces a structural presumption for mergers that "Entrench or Extend a Dominant Position."^{70,71} The MGs explain that such mergers may not only offend Section 7 of the Clayton Act but may also offend Section 2 of the Sherman Act.⁷² What type of mergers are behind this guideline?

In Guideline 6, the Agencies aim for mergers that *entrench market power*. The basic principle

⁶⁶MGs, p. 13. See Shapiro (2021) for a discussion on quantifying the raising rivals' costs effect in a bargaining setting for the 2018 AT&T and Time Warner merger that was challenged by the Department of Justice. Rogerson (2020) and Rogerson (2021) incorporate bargaining between upstream and downstream firms to investigate the competitive effect of vertical mergers. Rogerson (2020) shows that a vertical merger allows the merged firm to increase the price that it charges rival downstream firms for inputs by increasing its bargaining leverage over these downstream rivals. Donna et al. (2024) estimate a structural bargaining model to quantify the bargaining leverage effect and discuss the implication for vertical mergers.

 69 Nocke and Whinston (2010) study the dynamic horizontal merger review problem by the Agencies. 70 Ibid, p. 18.

⁶²Emphasis added; Ibid., p. 16.

⁶³Ibid., p. 13.

⁶⁴Federal Trade Commission Complaint (2022). A theoretical model and a discussion are provided in Alderman, Blair and Donna (2024).

⁶⁵MGs, p. 13. "A merger that gives the merged firm increased visibility into competitively sensitive information could undermine rivals' ability or incentive to compete aggressively or could facilitate coordination." (MGs, p. 17.) See U.S. Department of Justice v. Agri Stats Inc. (2023) and the discussion by Donna and Walsh (2023). Brown and MacKay (2023) study the implications for competition of pricing algorithms; Cho and Williams (2024) build an algorithmic pricing model that generates collusive outcomes after shutting down all explicit and implicit collusion channels; see also the references therein.

⁶⁷MGs, p. 13.

⁶⁸Ibid., pp. 15-16.

⁷¹Related literature includes the study of "killer acquisitions" (*e.g.*, Kamepalli, Rajan and Zingales 2020; Cunningham, Ederer and Ma 2021; Cabral 2021; and Letina, Schmutzler and Seibel 2021), and the study of the effect of switching cost on elasticities (*e.g.*, Donna 2021) and competition (*e.g.*, Klemperer 1987*a*, Klemperer 1987*b*, Shum 2004, Farrell and Klemperer 2007, Dubé, Hitsch and Rossi 2009, Dubé, Hitsch and Rossi 2010, Shcherbakov 2016, and Hortaçsu, Madanizadeh and Puller 2017).

⁷²United States v. Grinnell Corp., 384 U.S. 563 (1966).

is that a dominant firm may extend its market power through mergers and acquisitions. The Agencies may use this presumption to enforce Section 7 or Section 2 violations in non-horizontal mergers that involve complementary-product or product-extension mergers.⁷³

Imagine that we are at the year-end of 2022. Google tries to develop its own chatbot with artificial intelligence (AI) capabilities and fails.⁷⁴ Now Google wants to buy the AI chatbot company Anthropic. Will the Agencies challenge Google's acquisition of Anthropic? The answer is likely "yes" on the basis of Guideline 6, because the merger will entrench Google's market power in this market. However, this is a non-horizontal merger.

Examples of Guideline 6 application include: The previously cited United States v. Grinnell Corp., 384 U.S. 563 (1966); Fruehauf Corp. v. FTC, 603 F.2d 345 (2d Cir. 1979); and U.S. v. Microsoft Corp., 253 F.3d 34 (D.C. Cir. 2001).

Guideline 7 states that "the Agencies Consider Whether [a Trend Toward Consolidation] Increases the Risk a Merger May Substantially Lessen Competition or Tend to Create a Monopoly."^{75,76} This guideline states that the *trend* toward vertical integration can be used as a structural presumption in non-horizontal mergers due to "implications for the *competitive dynamics* of the industry."⁷⁷

The trend toward consolidation widens the concerns in Guideline $5.^{78}$ The trend may also increase the bargaining leverage of the merging firms and thereby increase the risks that are discussed in Guidelines 1-6.⁷⁹

Examples of Guideline 7 application include: Brown Shoe Co. v. U.S., 370 U.S. 294 (1962); United States v. Pabst Brewing Co., 384 U.S. 546 (1966); and United States v. Von's Grocery Co., 384 U.S. 270 (1966).

Guideline 8 is also concerned with dynamic considerations when it states that "When a Merger is Part of a Series of Multiple Acquisitions, the Agencies May Examine the Whole Series."^{80,81} Guideline 8 can be used as an anticompetitive presumption in non-horizontal mergers.

Consider Google's acquisition activity during the years that preceded the DOJ's 2020 lawsuit.⁸² Bet, Blair and Donna (2022) report that "Google merger-and-acquisition activity was hectic during the years prior to the DOJ lawsuit. Between 2003 and 2018, Google engaged in 225 mergers and acquisitions, an average of 15 per year."⁸³ Google's acquisition of nascent

⁷³For a recent example, see the 2020 DOJ lawsuit against Google, which alleges a §2 Sherman Act violation. See Bet, Blair and Donna (2022) for a discussion.

 $^{^{74}}$ For a recount of the AI race inside Silicon Valley, see Weise et al. (2023).

 $^{^{75}}MGs$, p. 22.

⁷⁶Related papers include, *e.g.*, Ashenfelter, Hosken and Weinberg (2015), Wollmann (2019), Wollmann (2020), and Prager and Schmitt (2021).

⁷⁷Emphasis added; MGs, p. 22.

 $^{^{78}}$ Ibid.

⁷⁹See Donna et al. (2024, Section 6).

⁸⁰Ibid., p. 23.

⁸¹For references to the economic literature, see footnotes 32, 38, and 71.

⁸²United States of America et al. v. Google LLC (2020).

⁸³Bet, Blair and Donna (2022, p. 32).

vertical search providers may foreclose competition and create barriers to entry.⁸⁴ These are not horizontal mergers but may be challenged using Guideline 8.

Guideline 9, about platforms, can be used as a structural presumption in non-horizontal "[m]ergers involving platforms [...], even when a platform merges with a firm that is neither a direct competitor nor in a traditional vertical relationship with the platform."^{85,86} For example, a platform that offers its own products on the platform may favor them to the detriment of the rivals' product "with rivals of the advantaged product *exiting* or becoming less attractive."⁸⁷

Some anticompetitive practices may include self-preferencing; misappropriation of vertical search content; use of scale as a barrier to entry; and anticompetitive exclusivity agreements.⁸⁸ In a platform setting, switching costs, network effects, and scale economies may exacerbate the effects of anticompetitive practices.

There have been many antitrust cases that involve platforms in recent years. Examples include: The mentioned lawsuits against Google⁸⁹ and Facebook;⁹⁰ the Amazon lawsuit;⁹¹ the challenge of Sabre's acquisition of Farelogix (digital platforms that provide booking services to airlines);⁹² the complaints against Ticketmaster and Live Nation;⁹³ the concerns with regard to rideshare platforms and the gig economy;⁹⁴ and the complaint against Visa and Mastercard from Block (formerly Square);⁹⁵ inter alia. These examples show how Guideline 9 may be combined with Guidelines 1-6 depending on the industry specifics.⁹⁶

Guideline 10 states that "When a Merger Involves Competing Buyers, the Agencies Examine Whether It May Substantially Lessen Competition for Workers, Creators, Suppliers, or

⁸⁷Emphasis added; MGs, footnote 48.

 $^{^{84}}$ Ibid.

⁸⁵MGs, p. 23.

⁸⁶For recent studies about platforms and two-sided markets concerning market power, competition policy, anticompetitive practices, and regulation see, *e.g.*, Hovenkamp (2018), De Corniere and Taylor (2019), Bet, Blair and Donna (2022), Reimers and Waldfogel (2023), Waldfogel (2024), and the references therein. See Rysman (2009) and Jullien, Pavan and Rysman (2021) for surveys.

⁸⁸For definitions and a discussion see Bet, Blair and Donna (2022, Section IV).

⁸⁹ United States of America et al. v. Google LLC (2020) and the related discussion in Bet, Blair and Donna (2022). Athey and Scott Morton (2022, p. 703) recommend an anticompetitive presumption regarding platform annexation by a dominant firm, where they define *platform annexation* as the situation where a "platform annexes multi-homing tools and manipulates them to reduce multi-homing and competition." Hovenkamp (2023) develops a competition model between two-sided search platforms with a dominant player.

⁹⁰Federal Trade Commission v. Facebook (2021).

⁹¹Federal Trade Commission and Plaintiff States v. Amazon.com, Inc. (2023).

⁹²United States v. Sabre Corporation, Sabre Glbl Inc., Farelogix, Inc., and Sandler Capital Partners V, L.P. (2019).

 $^{^{93}}$ See Andrzejewski (2023) for a summary.

 $^{^{94}}$ See Peterson and Steinbaum (2023) for a discussion.

⁹⁵Block, Inc. v. Visa Inc. et al., U.S., Eastern District of New York, case 1:23-cv-05377.

⁹⁶Ohio v. American Express Co., 585 U.S. (2018) provides an example for "platforms with the "key feature ... that they cannot make a sale to one side of the platform without simultaneously making a sale to the other." MGs, footnote 47 citing Ohio v. American Express Co., 138 S. Ct. 2274, 2280 (2018).

Other Providers."^{97,98} Guideline 10 is among the most notable antitrust developments in the MGs.⁹⁹ It introduces a structural presumption in buyer markets by saying that "a merger's harm to competition among buyers [notably, employers in labor markets] is not saved by benefits to competition among sellers."¹⁰⁰

This structural presumption may apply to non-horizontal markets when the merger increases bargaining leverage in the vertical market or when combined with other guidelines; for example, with Guideline 5 (a merger of employers that may limit access to products that are used by the rivals) or Guideline 7 (a merger between buyers that displays a trend toward vertical integration).

Examples include: Mandeville Island Farms v. American Crystal Sugar, 334 U.S. 219 (1948); and the recent cases of Nat'l Collegiate Athletic Ass'n v. Alston, 141 S. Ct. 2141, 210 L. Ed. 2d 314 (2021); and U.S. v. Bertelsmann SE & CO. KGaA, et al. (2022): the Penguin Random House's proposed acquisition of Simon & Schuster, which was blocked by the District Court.¹⁰¹

Guideline 11 states that "When an Acquisition Involves Partial Ownership or Minority Interests, the Agencies Examine Its Impact on Competition."^{102,103} This guideline refers to partial mergers that do not completely eliminate the horizontal rival but may nevertheless cause horizontal harm. It introduces an anticompetitive presumption in non-horizontal mergers for "both cross-ownership [...] as well as common ownership" partial acquisitions that may substantially lessen competition using, *e.g.*, the presumptions in any of the previous guidelines.¹⁰⁴

Crucially, the "Agencies recognize that cross-ownership and common ownership can reduce competition by softening firms' incentives to compete, *even absent any specific anticompetitive act or intent.*"¹⁰⁵ The emphasized novel part is backed up by the mentioned academic literature.

Examples may include: Common ownership and coordination in the U.S. airline industry (Azar, Schmalz and Tecu 2018, Park and Seo 2019, and Bet 2021); the MillerCoors joint venture in the U.S. beer industry (Miller and Weinberg 2017); and the merger request of Spirit Super

⁹⁷MGs, p. 26.

⁹⁸For investigations about market power, competition policy, antitrust, and mergers in labor markets see, *e.g.*, Naidu (2010), Naidu, Nyarko and Wang (2016), Naidu, Posner and Weyl (2018), Azar, Marinescu and Steinbaum (2022), Naidu and Posner (2022), Marinescu and Hovenkamp (2019), Azar et al. (2020), Azar, Berry and Marinescu (2022), Bassier, Dube and Naidu (2022), Berger et al. (2023), Naidu, Nyarko and Wang (2023), and the references therein. See also Hemphill and Rose (2017).

⁹⁹A summary and some prominent cases are discussed by ProMarket Writers (2024).

¹⁰⁰MGs, p. 27.

¹⁰¹United States v. Bertelsmann Se & Co. Kgaa, Penguin Random House, Llc, Viacomcbs, Inc., and Simon & Schuster, Inc. (2022). See Athey et al. (2023) for discussions.

 $^{^{102}}MGs$, p. 28.

¹⁰³The literature that investigates the competitive effects of common ownership is relatively recent and has experienced an explosion in recent years. An early investigation in the airline industry is provided by Azar, Schmalz and Tecu (2018). Antón et al. (2023) "show that managerial incentives can serve as a mechanism that connects common ownership to softer competition." Antón et al. (2024) investigate theoretically and empirically the effect of common ownership on innovation. There are several surveys that focus on economics and finance (Schmalz 2018; Schmalz 2021), industrial organization (Backus, Conlon and Sinkinson 2019), and law and economics (Elhauge 2020), inter alia.

 $^{^{104}}$ See OECD (2017) and Moss (2020) for discussions and examples.

¹⁰⁵Emphasis added; MGs, p. 28.

and Palisade Investment Partners Consortium, which was withdrawn after the concerns released by the Australian Competition and Consumer Commission (ACCC).¹⁰⁶ Additional examples include United States v. E. I. du Pont de Nemours & Co., 353 U.S. 586 (1957), Denver & Rio Grande Western R. Co. v. United States, 387 U.S. 485 (1967), and U.S. v. Dairy Farmers of America, Inc., 426 F.3d 850 (6th Cir. 2005).

In sum, the anticompetitive presumptions in the MGs are consistent with the latest developments from the economics literature, and in particular with respect to the theories of harm in non-horizontal markets. The MGs pave the way to challenge mergers that may pose anticompetitive concerns in markets that involve platforms, digital ecosystems, labor markets, common ownership, and rivals' exit that might substantially lessen competition.

3 A Path Forward

The remainder of this article discusses the economic rationale for a structural presumption in non-horizontal mergers that may cause rivals' exit due to the exercise of market power. We proceed in three steps. In this section, we summarize the exit concerns in the MGs and present the terminology. Section 4 presents a model that is used to discuss the economic rationale for the anticompetitive presumptions. The structural presumptions are discussed in Section 5.

3.1 Exit concerns in the 2023 Merger Guidelines

The (withdrawn) VMGs expressed exit concerns in only one instance.¹⁰⁷ By contrast, the MGs incorporate several anticompetitive presumptions and concerns with regard to mergers that may cause rivals' exit, consistent with Donna and Pereira (2023*a*), Donna and Pereira (2023*b*), and Donna and Pereira (2023*c*). A summary is available in Appendix A.

¹⁰⁶The ACCC website notes: "The ACCC was concerned that the proposed acquisition may substantially lessen competition in the supply of port services for long-term bulk cargo customers in Victoria by reducing competition between the Port of Portland and Port of Geelong. [...] "We were concerned the common fund management and ownership interests between the Port of Geelong and the Port of Portland would reduce competition for customers between the ports over the medium to long term," ACCC Chair Gina Cass-Gottlieb said." (Emphasis added; Australian Competition and Consumer Commission 2022.) We thank Martin Schmalz for the pointer.

¹⁰⁷"In identifying whether a vertical merger may diminish competition due to unilateral foreclosure or raising rivals' costs, the Agencies generally consider whether the following conditions are satisfied: (1) Ability: By altering the terms by which it provides a related product to one or more of its rivals, the merged firm would likely be able to cause those rivals (a) to lose significant sales in the relevant market (for example, *if they are forced out of the market*; if they are deterred from innovation, entry, or expansion, or cannot finance those activities; or if they have incentives to pass on higher costs through higher prices) or (b) to otherwise compete less aggressively for customers' business." (Emphasis added; VMGs, §4.a.)

3.2 Terminology

We now discuss the economic rationale for these structural presumptions. We focus on concentrated industries where firms sell differentiated products.¹⁰⁸ We use the expressions *merger* and *integration* interchangeably. Two products are (gross) substitutes if the increase in the price of one of them raises the demand for the other. Two products are (gross) complements if the increase in the price of one of them reduces the demand for the other.

We refer to the activity level that is closest to final consumers as the *downstream level* the level of distribution or retail—and to the activity level furthest from final consumers as the *upstream level*—the level of supply or manufacture. We refer to an upstream firm as a *manufacturer* and a downstream firm as a *retailer*.¹⁰⁹ The product of a manufacturer is a manufactured product, and the product of a retail firm is a retail product. We adopt the corresponding terminology for the other elements of industry activity, such as price, marginal cost, and profit. We define a *product* as a combination of manufacturer and retailer. Hence, products of a manufacturer that are sold through two distinct retailers are two different manufactured products, and the products of two distinct manufacturers that are sold by the same retailer are two different retail products.¹¹⁰

A horizontal merger combines firms that compete at the same level of production or distribution. A vertical merger combines firms that operate at different production and distribution levels.¹¹¹ Firms are rivals if they produce substitute products.¹¹²

The increase in the price of a product decreases the sales of that product and increases the sales of the alternative products, as consumers substitute their purchases from the former product. We employ the term *diversion of sales* to describe the replacement of the sales of a product by sales of the substitute products that is caused by the increase in the product's own price.¹¹³ *Market power* is the ability to mantain prices profitably above marginal costs. *Unilateral effects* are the exercise of market power that is enabled by a merger when firms act independently of their rivals.

Below, we make several statements about whether a merger might enable the exercise of market power. These statements refer to the sign or direction of the effect. To establish harm, it is also necessary to show the risk of the likely substantial magnitude of the effect. Merger

¹⁰⁸Competition problems usually arise in concentrated industries. Most of our analysis applies regardless of whether products are homogeneous or differentiated. However, focusing on the latter simplifies the exposition. In most concentrated industries, brands matter.

¹⁰⁹Alternatively, a manufacturer is a firm that produces a production factor, and a retailer is a firm that uses a manufactured product to produce a final product.

¹¹⁰For example, the same brand of soda that is sold through two distinct supermarket chains are two different manufactured products, and two distinct brands of soda that are sold by the same supermarket chain are two different retail products.

¹¹¹Alternatively, a horizontal merger combines firms that produce substitute products, and a vertical merger combines firms that produce complementary products.

¹¹²Alternatively, firms are rivals if they operate in the same relevant market.

¹¹³For example, the increase in the price of Coca-Cola reduces its demand because some consumers stop purchasing cola beverages altogether, while others switch to alternative cola brands, such as Pepsi-Cola. Thus, a fraction of the decreased demand for Coca-Cola switches to Pepsi, a substitute product for Coca-Cola.

evaluation involves considering the facts of the industry, such as the relative size of the firms and diversion ratios, or, when possible, the use of computational models. The exercise of market power and the efficiencies that are induced by a merger may affect prices, quality, product variety, and innovation.

To simplify the exposition, we focus exclusively on prices. However, the conclusions extend to other measures of performance.

The following assumptions provide the basis of our theories of benefit and of harm.

(A1) Before and after the merger, the manufacturer of the merged firm supplies products to the retailer of the merged firm.

(A2) The merged firms sell manufactured products to rival retailers.

(A3) The merged firms buy manufactured products from manufacture rivals.

Assumption (A1) is necessary for a vertical merger to enable strategic efficiencies. Assumptions (A2) and (A3) are necessary for a vertical merger to cause harm, *i.e.*, enable the exercise of market power, upstream and downstream, respectively.

4 Model

4.1 Setup

This section presents a model that is used to discuss the economic rationale for the anticompetitive presumptions in Section 5. The model includes the basic elements of the analysis in the simplest possible setting, so that the discussion can focus on policy.¹¹⁴

Let x_i , $i = 1, \dots, N$, be some real variables. Denote the $N \times 1$ vector by $\mathbf{x} := (x_1, \dots, x_N)'$. Consider the differentiable function $f(\mathbf{x}): \mathbb{R}^J \to \mathbb{R}$. Let $\nabla_x f := \frac{\partial f(\mathbf{x})}{\partial \mathbf{x}'}$ be the $J \times J$ Jacobian matrix of $f(\cdot)$ with respect to \mathbf{x} . Let \odot be the Hadamard product.

Consider an industry with a two-layered vertical structure: The manufacture and retail layers. In the manufacturer layer, manufacturers produce manufactured products, which they sell to firms in the retail layer. In the retail layer, firms called retailers use manufactured products to produce retail products that they sell to final consumers. Let v denote manufacture and r denote retail.

Let v_j be manufactured product j, and r_j be retail product j. Manufactured product v_j is used to produce retail product r_j . Let \mathcal{J}^{τ} be the set of type $\tau = v, r$ products. There are two manufactured products $\mathcal{J}^v = \{v_1, v_2\}$, and two retail products $\mathcal{J}^r = \{r_1, r_2\}$. Let \mathcal{F} be the set of firms. There are 4 firms: $\mathcal{F} = \{f_1, \cdots, f_4\}$. The firms own the manufactured and retail products.

Let \mathcal{J}_{f}^{τ} be the set of type $\tau = v, r$ products of firm f. We defer the presentation of particular property structures until Section 5. Let $\Omega^{\tau\tau'}$ be a matrix with generic element $\{\omega_{jj'}^{\tau\tau'}\}$, where

¹¹⁴For additional details about the model, see Donna and Pereira (2023a) and Donna, Pereira and Pu (2024).

 $j \in \mathcal{J}^{\tau}$ and $j' \in \mathcal{J}^{\tau'}$, with $\tau, \tau' = v, r$, and:

$$\omega_{jj'}^{\tau\tau'} = \begin{cases} 0 \iff j \in \mathcal{J}_f^{\tau}, \, j' \in \mathcal{J}_{f'}^{\tau'}, \, f' \neq f \\ 1 \iff j \in \mathcal{J}_f^{\tau}, \, j' \in \mathcal{J}_{f'}^{\tau'}, \, f' = f \end{cases}$$

Matrices $\Omega := \{\Omega^{\tau\tau'} | \tau, \tau' = \upsilon, r\}$ represent the property structure of the industry. Let \mathbb{I}_N be an N-dimensional identity matrix, and \mathbb{O}_N be an N-dimensional square all-zeros matrix.

For j = 1, 2, let p_j be the price of retail product j, w_j be the price of manufactured product j, and $c_j = \theta$ be the marginal cost of manufactured product j.¹¹⁵ Without loss of generality, the only marginal cost that is associated with the production of a retail product is the price of the associated manufactured product.¹¹⁶ For j = 1, 2 and $\mathbf{p} \in (0, +\infty)^2$, $D_j(\mathbf{p})$ is the demand of retail product j. Let function $D_j(\cdot)$ be twice continuously differentiable. Let $\varphi_f \in (0, +\infty)$ be the fixed cost of firm f.

Consider the following assumptions:

(A4a) Retail products have downward-sloping demand curves.

(A4b) Retail products are (gross) substitutes.

(A4c) The demand for retail products is log-concave.¹¹⁷

(A4c') The slopes of the demand retail functions are increasing in the other retail prices.¹¹⁸

Assumption (A4a) means that retail firms have market power, a natural occurrence in concentrated markets. Assumption (A4b) means that retailers compete with each other. We defer the discussion of assumptions (A4c) and (A4c') until Section 4.2.

Let $\overline{\theta}$ and \overline{p} be such that $0 < \overline{\theta} < \overline{p} < \infty$. Let $(\mathbf{w}, \mathbf{p}) \in [\overline{\theta}, \overline{p}]^4$. The profit of r_j is:

$$\pi_j^r(\mathbf{p}; w_j) := (p_j - w_j) D_j(\mathbf{p});$$

the profit of v_j is:

$$\pi_j^v := (w_j - c_j) D_j;$$

and the profit of a firm f is:

$$\Pi_f := \sum_{\tau=\upsilon,r} \sum_{j\in\mathcal{J}_f^\tau} \pi_j^\tau - \varphi_f.$$

The game unfolds in two stages: In stage 1, the owners of manufactured products choose their prices. In stage 2, the owners of retail products choose their prices.¹¹⁹ At the end of each stage, all firms observe the actions taken. The equilibrium concept is subgame perfection.

¹¹⁶Other possible types of retail marginal costs play no role in our analysis.

¹¹⁷Each product's own-price elasticity of demand decreases with the prices of the other products.

¹¹⁸For $j \neq j', \frac{\partial^2 D_j}{\partial p_{j'} \partial p_j} > 0.$

 $^{^{115}\}mathrm{The}$ symmetry of marginal costs only simplifies exposition.

¹¹⁹In other words, at each stage, firms play a Bertrand-Nash game.

4.2 Equilibrium

In this subsection, we compute the equilibrium by backward induction. We present the equilibrium conditions in a way that encompasses all possible property structures. Denote equilibrium values with a superscript "*".

4.2.1 Second-Stage Game: Retail Price Competition

In matrix form, the equilibrium conditions for **p** are:

$$\Phi^{r}(\mathbf{p}^{*};\mathbf{w},\Omega) := \mathbf{D}(\mathbf{p}^{*}) + \left(\Omega^{rr} \odot [\nabla_{p}D]'\right)(\mathbf{p}^{*} - \mathbf{w}) + \left(\Omega^{rv} \odot [\nabla_{p}D]'\right)(\mathbf{w} - \mathbf{c}) = 0.$$
(1)

Equation (1) defines, implicitly, the best-response function:

$$\mathbf{p} = \mathbf{B}^r(\mathbf{p}, \mathbf{w}) = \left(egin{array}{c} b_1^r\left(p_2; \mathbf{w}
ight) \ b_2^r\left(p_1; \mathbf{w}
ight) \end{array}
ight).$$

Best-response function $b_j^r(\cdot)$ is increasing in $p_{j'}, j' \neq j$. Retail prices are strategic complements.¹²⁰ Assume that:¹²¹

$$\frac{\partial b_j^r}{\partial p_{j'}} < 1, \quad j' \neq j. \tag{2}$$

Then, the second-stage game has a Nash equilibrium in pure strategies.¹²² Denote it by:

$$\mathbf{p}^* = \mathbf{P}(\mathbf{w}) = \begin{pmatrix} p_1^*(\mathbf{w}) \\ p_2^*(\mathbf{w}) \end{pmatrix} = \begin{pmatrix} b_1^r(p_2^*; \mathbf{w}) \\ b_2^r(p_1^*; \mathbf{w}) \end{pmatrix}$$

4.2.2 First-Stage Game: Manufacturer Price Competition

Given $\mathbf{P}(\mathbf{w})$, let $\widetilde{D}_j(\mathbf{w}) := D_j(\mathbf{p}^*)$, and call $\widetilde{D}_j(\cdot)$ the demand function of v_j . Consider the following assumption:

(A5a) Each manufactured product has a downward-sloping demand curve.

Let:

$$\widetilde{\pi}_{j}^{v}(\mathbf{w};c_{j}) := (w_{j} - c_{j})\widetilde{D}_{j}(\mathbf{w}),$$
$$\widetilde{\pi}_{j}^{r}(\mathbf{w}) := (p_{j}^{*}(\mathbf{w}) - w_{j})\widetilde{D}_{j}(\mathbf{w}),$$

¹²⁰This follows from assumptions (A4b) and (A4c) for some property structures, and from assumptions (A4b) and (A4c') for other property structures. For the concept of strategic substitutes and strategic complements, see *e.g.*, Bulow, Geanakoplos and Klemperer (1985).

¹²¹This follows from standard stability conditions.

 $^{^{122}}$ Existence follows from, *e.g.*, Topkis (1998, Theorem 4.2.1). The equilibrium may not be globally unique. However, given enough differentiability, it is locally unique. See Debreu (1970).

and

$$\widetilde{\Pi}_f := \sum_{\tau=\upsilon,r} \sum_{j \in \mathcal{J}_f^\tau} \widetilde{\pi}_j^\tau - \varphi_f.$$

Let:

$$\frac{d\widetilde{D}_{j}(\mathbf{w})}{dw_{j'}} := \sum_{k} \frac{\partial D_{j}}{\partial p_{k}} \frac{p_{k}^{*}}{\partial w_{j'}}$$

Let: $\overline{\mathbf{P}} := \mathbf{P} - \mathbf{w}$. In matrix form, the equilibrium conditions for \mathbf{w} are:¹²³

$$\Phi^{\upsilon}(\mathbf{w}^*;\Omega) = \widetilde{\mathbf{D}}(\mathbf{w}^*) + \left(\Omega^{\upsilon\upsilon} \odot \left[\nabla_w \widetilde{\mathbf{D}}\right]'\right)(\mathbf{w}^* - \mathbf{c}) + \left(\Omega^{\upsilon r} \odot \left[\nabla_w \widetilde{\mathbf{P}}\right]'\right)\widetilde{\mathbf{D}} + \left(\Omega^{\upsilon r} \odot \left[\nabla_w \widetilde{\mathbf{D}}\right]'\right)(\mathbf{P} - \mathbf{w}^*) = 0.$$
(3)

Equation (3) defines, implicitly, the best-response function:

$$\mathbf{w} = \mathbf{B}^{\upsilon}(\mathbf{w}, \mathbf{c}) = \left(egin{array}{c} b_1^{\upsilon}\left(w_2; \mathbf{c}
ight) \ b_2^{\upsilon}\left(w_1; \mathbf{c}
ight) \end{array}
ight).$$

Then, an equilibrium exists.¹²⁴ Denote it by:

$$\mathbf{w}^* = \mathbf{W}(\mathbf{c}) = \begin{pmatrix} w_1^*(\mathbf{c}) \\ w_2^*(\mathbf{c}) \end{pmatrix} = \begin{pmatrix} b_1^v(w_2^*;\mathbf{c}) \\ b_2^v(w_1^*;\mathbf{c}) \end{pmatrix}.$$

A subgame-perfect equilibrium is: $(w^*, p^*) = (\mathbf{W}(\mathbf{c}), \mathbf{P}(\mathbf{W}(\mathbf{c}))).$

5 Theories of Benefit and Harm

This section discusses the *benefits* and *harms* of vertical mergers: Social-welfare increasing and diminishing effects, respectively.¹²⁵ Appendix B presents a complementary case.

5.1 Backdrop

We start by presenting a version of the model that is neutral in terms of efficiencies and market power: Assumptions (A1)-(A3) do not hold. This case will serve as a reference.

Consider the model of Section 4: Let firms f_1 and f_2 own manufactured products v_1 and

¹²³Where $\nabla_w \widetilde{\mathbf{D}} := \nabla_p \mathbf{D} \nabla_w \mathbf{P}$.

¹²⁴Existence follows from standard arguments given continuity and compactness; see, *e.g.*, Fudenberg and Tirole (1991, Theorem 1.1). The conditions for $\widetilde{\mathbf{D}}(\cdot)$ to be log-concave are more stringent than those for $\mathbf{D}(\cdot)$. Furthermore, conditions that ensure log-concavity of $\widetilde{\mathbf{D}}(\cdot)$ for some property structures my fail for others.

¹²⁵The discussion in this section involves direct and indirect effects. Given our focus on policy, we restrict attention to the direct, dominating effects. See Donna and Pereira (2023a) and Donna, Pereira and Pu (2024) for details.

 v_2 , respectively, and firms f_3 and f_4 own retail products r_1 and r_2 , respectively.¹²⁶ Figure 1 represents the situation. Note that v_1 is sold only to r_1 and v_2 is sold only to r_2 . Denote equilibrium values by superscript "s".

For $f = f_3$, f_4 and j = 1, 2, the equilibrium conditions for p_j are:

$$\frac{\partial \Pi_f}{\partial p_j} = \underbrace{D_j}_{\text{margin effect}>0} + \underbrace{(p_j^s - w_j)\frac{\partial D_j}{\partial p_j}}_{\text{volume-of-sales effect} < 0} = 0.$$
(4)

The p_j equilibrium condition balances two effects: First is the margin effect: Given the quantity sold of r_j , an increase in p_j increases the margin at which each unit of r_j is sold. This effect increases profit. Second is the volume-of-sales effect: An increase in p_j decreases the quantity demanded of r_j . This effect reduces profit. These two effects can be interpreted as the marginal benefit and the marginal cost of raising p_j , respectively.

For $f = f_1, f_2$ and j = 1, 2, the equilibrium conditions for w_j are:

$$\frac{\partial \widetilde{\Pi}_{f}}{\partial w_{j}} = \underbrace{\widetilde{D}_{j}}_{\text{margin effect}>0} + \underbrace{(w_{j}^{s} - c_{j})\frac{d\widetilde{D}_{j}}{dw_{j}}}_{\text{volume-of-sales effect} < 0} = 0.$$
(5)

These equilibrium conditions have a similar interpretation as those for p_j . Denote by $(\mathbf{w}^s(\mathbf{c}), \mathbf{p}^s(\mathbf{c}))$ the subgame-perfect equilibrium for this case.

5.2 Benefits

A merger may generate efficiencies that result in lower marginal costs, and thereby in lower prices and higher social welfare.

5.2.1 Two Types of Benefits

A vertical merger combines complementary assets at different levels of production and distribution. This feature may enable two types of efficiencies: technological and strategic. We discuss them in turn.¹²⁷

Technological efficiencies refer to technology—or organizational—induced efficiencies. A vertical merger might eliminate contractual frictions, improve communication flows, and align incentives within the merged firm, which facilitate the rationalization of investments, inventory management, and production. Consequently, marginal costs might fall. See, *e.g.*, Williamson (1968) and Farrell and Shapiro (2000).¹²⁸

¹²⁶Hence, if $\tau = v, r$, then $\Omega^{\tau\tau} = \mathbb{I}_2$, and if $\tau \neq \tau'; \tau, \tau' = v, r$, then $\Omega^{\tau\tau'} = \mathbb{O}_2$.

 $^{^{127}\}mathrm{The}$ MGs discuss procompetitive efficiencies in the rebuttal Subsection 3.3.

 $^{^{128}}$ A system of quotas, interpreted as a vertical integration, ameliorates transaction costs and could be more efficient than a vertical market in the presence of frictions, as is shown by Donna and Espín-Sánchez (2024).

Strategic efficiencies refer to efficiencies that are induced by strategic behavior. Before the merger, the manufacturer and the retailer of the merged firm maximize their profits independently. If there is market power upstream and downstream, both firms charge prices above their marginal costs. After the merger, the merged firm maximizes the joint profit of its manufactured and retail products. Given assumption (A1), it appropriates the increase in retail sales that is caused by a decrease in the manufacture's price. Thus, the merged firm sets the manufacturer price that is charged to its retail unit at the manufacture's marginal cost: The merged firm eliminates the manufacture's margin on these internal "sales." The decrease in the manufacture's price (the retail marginal cost) leads the merged firm to reduce the retail price. The elimination of the manufacture margin is usually referred to as the *elimination of double marginalization* (EDM). See, *e.g.*, Cournot (1838) and Spengler (1950).

Both horizontal and vertical mergers may generate technological efficiencies. Only vertical mergers generate strategic efficiencies.

5.2.2 Vertical Merger and Efficiency Benefits

Next, we present a version of the model that allows technological and strategic efficiencies but enables no exercise of market power: Assumption (A1) holds; but assumptions (A2) and (A3) do not hold.

Consider the model of Section 4: Let firm f_1 own products $\{v_1, r_1\}$; firm f_2 owns product v_2 ; firm f_4 owns product r_2 ; and firm f_3 be inactive.¹²⁹ The merged (integrated) firm— f_1 —owns two complements. See Figure 2. Note again that v_1 is sold only to r_1 and v_2 is sold only to r_2 . Denote equilibrium values by superscript "e".

For $f = f_4$, the equilibrium condition for p_2 is as in equation (4). For $f = f_1$, the equilibrium condition for p_1 is:¹³⁰

$$\frac{\partial \Pi_{f_1}}{\partial p_1} = \underbrace{D_1}_{\text{margin effect}>0} + \underbrace{(p_1^e - c_1)\frac{\partial D_1}{\partial p_1}}_{\text{volume-of-sales effect} < 0} = 0.$$

For f_2 , the equilibrium condition for w_2 is as in equation (5). For f_1 , $w_1^e = c_1 = \theta$. Denote by $((c_1, w_2^e(\mathbf{c})), (p_1^e(\mathbf{c}), p_2^e(\mathbf{c})))$ the subgame-perfect equilibrium for this case.

Firm f_1 —a vertically integrated firm—takes the manufacturer's marginal cost— c_1 —as its opportunity cost. It is equivalent to setting the manufacturer's price at the marginal cost: $w_1^e = c_1 < w_1^s$.¹³¹ The decrease in w_1 – the marginal cost of r_1 – due to the EDM causes a decrease in the price of r_1 : $p_1^e < p_1^s$. Since retail products are strategic complements, firm f_4 responds by also reducing its retail price, but by less than the merged firm: $p_1^e < p_2^e < p_2^s$, where the first inequality follows from equation (2).

In turn, this leads firm f_2 to decrease its manufactured price, but by less than does the

¹²⁹Hence, if $\tau = \tau', \tau, \tau' = \upsilon, r$, then $\Omega^{\tau\tau'} = \mathbb{I}_2$; and if $\tau \neq \tau', \tau, \tau' = \upsilon, r$, then $\Omega^{\tau\tau'} = (1.1)' \cdot (1, 0)$.

¹³⁰Now, retail prices are strategic complements if (A4b) and (A4c) hold.

 $^{^{131}}$ The inequality follows from assumption (A5a).

merged firm: $\theta = w_1^e < w_2^e < w_2^s$. Thus, if the firms that own v_1 and r_1 merge, due to the EDM, manufacturer and retail prices decrease: $\mathbf{w}^e < \mathbf{w}^s$ and $\mathbf{p}^e < \mathbf{p}^s$. Social welfare increases.

A vertical merger may also generate technological efficiencies, *i.e.*, reduce the merged firm's marginal cost by $\Delta_{\theta} \in (0, \theta]$. Then, $c_1^e = \theta - \Delta_{\theta} < c_1^s = \theta = c_2^e$. In this case, the price decreases and the resulting social welfare increase that is due to the merger will be larger than those that are generated by the EDM alone.

5.2.3 Challenges to Prove Benefits

Procompetitive efficiencies are credited in the merger review if they are merger-specific, verifiable, benefit consumers, and are not anticompetitive (MGs, Subsection 3.3). The EDM raises two issues with regard to whether it is a cognizable efficiency.¹³² First, assumption (A1) may not hold in a specific merger. In such a case, the merger involves no EDM (MGs, footnote 31).

Second, the EDM may be attainable contractually. For example, if the manufacture supplier and the retail client agree on a two-part tariff as a pricing schedule, where the variable price is set at the manufacturer's marginal cost, the EDM might be attained without the merger. Thus, the EDM might not be merger-specific.

However, even if the parties agree on a two-part tariff as a pricing schedule, it is unlikely that the pricing schedule will yield the efficient outcome that was described above. Information asymmetries and conflicting interests may prevent it.¹³³ Hence, even in the presence of such contractual arrangements, there might still be room for some EDM.

Whether a merger generates strategic efficiencies should be evaluated on case-by-case basis, grounded on the specificities of the industries, transactions, and agents that are involved—instead of being assumed *a priori*.

5.3 Harms

A merger may allow firms to cause price increases (directly or indirectly) that reduce social welfare.

5.3.1 Four Types of Harms

A vertical merger combines firms that operate at different production and distribution levels. This circumstance may give a vertically integrated firm the ability and incentive to engage in harmful conduct that is either unavailable or unprofitable for non-integrated firms. This conduct can be grouped into the *exercise market power*: The *induction of one or more rivals' exit*; the *foreclosure of rivals*; and *raising rivals' costs*. We will discuss them in turn.

With regard to the exercise of market power, let a firm own a manufacturer and a retailer

 $^{^{132}}$ For a broader discussion see see Kwoka and Slade (2019).

¹³³With regard to how information asymmetry might hinder efficient contracts between manufacturers and retailers, see, *e.g.*, Gal-Or (1991), Corbett, Zhou and Tang (2004), Wang, Lau and Lau (2012), and Tamayo and Tan (2021).

product—obtained possibly through a merger. If assumption (A2) ((A3)) holds, the increase in the firm's upstream (downstream) price causes downstream (upstream) a diversion of sales from the rivals' products to the firm's product. The profit of the firm's downstream (upstream) product increases. Hence, due to the diversion of retail (manufacture) sales, a manufacturer (retail) price increase—which might be unprofitable for an independent manufacturer (retailer)—may be profitable for a vertically integrated firm.

With regard to the induction of rivals' exit, the exercise of market power—which is enabled by a vertical merger—aside from directly reducing social welfare, also has an indirect negative effect on social welfare: A price increase diverts sales to substitute products. This *horizontal*-diversion of sales increases the profits of substitute products. Nevertheless, the exercise of market power that is caused by a horizontal merger shifts business *to* the merged firm's rivals. A price increase upstream (downstream) causes a diversion of sales from its downstream (upstream) complement to its substitute products. This *vertical*-diversion of sales decreases the profit of the downstream (upstream) complement.

Hence, if assumptions (A2) or (A3) hold, the exercise of market power that is caused by a vertical merger shifts business *from* the merged firm's rivals. This profit decrease may cause rivals to exit. Similarly, the efficiencies that are enabled by vertical mergers, which we discussed in Subsubsection 5.2.2, also divert sales *from* rivals.

With regard to the foreclosure of rivals, if assumption (A2) ((A3)) holds, the refusal by a firm to supply a manufactured (acquire a retail) product to a retailer (manufacturer) rival may foreclose that rival. The exit of the retail (manufacture) rival diverts sales to the firm's products downstream (upstream), whose profit increases. Hence, due to the vertical-diversion of retailer (manufacturer) sales, foreclosing a retailer (manufacturer), which might be unprofitable for an independent manufacturer (retailer), may be profitable for a vertically integrated firm. See, e.g., Hart and Tirole (1990), Ordover, Saloner and Salop (1990), Riordan and Salop (1995), and Salop (2018).

With regard to the raising rivals' costs, if assumption (A2) ((A3)) holds, it may be profitable for a vertically integrated firm to change the terms by which it transacts with retailer (manufacturer) rivals: Increase the price, or lower the quality, of the inputs that are supplied to rivals, which thereby raises rivals' costs (RRC). See, *e.g.*, Salop and Scheffman (1983) and Scheffman and Higgins (2014).

Input foreclosure is the limiting case of the exercise of market power upstream, whereby the merged firm sets a prohibitive manufacturer price.¹³⁴ With such a price—a price that no retail rival is willing to pay—the merged firm makes no sales of the manufactured product to the retail rivals. Similarly, customer foreclosure is the limiting case of the exercise of market power downstream, where the merged firm sets a prohibitive retail price for the rival's manufactured

 $^{^{134}}$ Input foreclosure refers to the situation where the merged firm refuses to sell a manufactured product to a retail rival. Consumer foreclosure refers to the situation where the merged firm refuses to acquire a manufactured product from a manufacturer rival. Formally, a price is prohibitive if it chokes off demand, *e.g.*, an infinite price.

product. With such a price, the merged firm makes no retail sales of the rival's manufactured product and, therefore, makes no purchases of the associated manufactured product.

In the remainder of this section, we focus on the two first types of harm, the exercise of market power and the induction of rivals' exit. For the two other types of harm—the foreclosure of rivals and raising rivals' costs—we refer the reader to the references that were indicated above.

5.3.2 Vertical Merger and Market Power Harm

Next, we present a version of the model that allows the exercise of market power but involves no EDM: Assumptions (A2) and (A3) hold; but assumption (A1) does not hold.

Consider the model of Section 4: Let firm f_1 own products $\{v_1, r_2\}$; firm f_2 owns product v_2 ; firm f_3 owns product r_1 ; and firm f_4 is inactive.¹³⁵ The merged firm— f_1 —owns a manufactured product and a retail product that is a substitute of the complement of its manufactured product. See Figure 3. Denote equilibrium values by superscript "m".

For $f = f_3$, the equilibrium condition for p_1 is as in equation (4). For $f = f_1$, the equilibrium condition for p_2 is:¹³⁶

$$\frac{\partial \Pi_{f_1}}{\partial p_2} = \underbrace{D_2}_{\text{margin effect}>0} + \underbrace{(p_2^m - w_2)\frac{\partial D_2}{\partial p_2}}_{\text{volume-of-sales effect}<0} + \underbrace{(w_1 - c_1)\frac{\partial D_1}{\partial p_2}}_{\text{vertical-diversion effect}>0} = 0$$

The equilibrium condition for p_2 includes the margin effect, the volume-of-sales effect, and an additional effect: The *vertical-diversion effect*.¹³⁷ An increase in p_2 diverts sales from r_2 to r_1 . It causes larger sales and profits for v_1 . Since firm f_1 owns both r_2 and v_1 , it appropriates this profit increase. Hence, the vertical-diversion effect decreases the marginal cost of increasing p_2 , which puts upward pressure on p_2 .

For $f = f_2$, the equilibrium condition for w_2 is as in equation (5). For $f = f_1$, the equilibrium condition for w_1 is:

$$\frac{\partial \widetilde{\Pi}_{f_1}}{\partial w_1} = \underbrace{\widetilde{D}_1}_{\text{margin effect}>0} + \underbrace{(w_1^m - c_1) \frac{d \widetilde{D}_1}{d w_1}}_{\text{volume-of-sales effect}<0} + \underbrace{\frac{\partial b_2^r}{\partial w_1} \widetilde{D}_2}_{\text{pass-through effect}>0} + \underbrace{(p_2^m - w_2^m) \frac{d \widetilde{D}_2}{d w_1}}_{\text{vertical-diversion effect}>0} = 0.$$

The equilibrium condition for w_1 includes the margin effect, the volume-of-sales effect, and ¹³⁵Hence, if $\tau = \tau', \tau, \tau' = v, r$, then $\Omega^{\tau\tau'} = \mathbb{I}_2$, furthermore:

$$\Omega^{\nu r} = \left(\begin{array}{cc} 0 & 1\\ 0 & 0 \end{array}\right) \quad \text{and} \quad \Omega^{r\nu} = \left(\begin{array}{cc} 0 & 0\\ 0 & 1 \end{array}\right).$$

 136 Now, retail prices are strategic complements if (A4b) and (A4c') hold. Condition (A4c') strengthens condition (A4c).

 137 This type of effect occurs in different contexts, upstream and downstream. To avoid terminology proliferation, we will refer to this set of effects as the *vertical-diversion effect*. Similarly, for the *horizontal-diversion effects*, in Appendix B. two additional effects: The third effect is the the vertical-diversion effect:¹³⁸ An increase in w_1 —the marginal cost of r_1 —increases p_1 . This diverts sales from r_1 to r_2 , which causes larger sales of and profits for r_2 . Since firm f_1 owns both v_1 and r_2 , it appropriates this profit increase. Hence, the vertical-diversion effect decreases the marginal cost of raising w_1 , which puts upward pressure on w_1 .

The fourth effect is the the pass-through effect. Given the quantity sold of r_2 , and that retail prices, $p_i^*(\cdot)$, are increasing in \mathbf{w}^{139} , an increase in w_1 raises p_2 . The margin at which each unit of r_2 is sold, and thereby the profit, increases. Since firm f_1 owns both v_1 and r_2 , the firm appropriates this profit increase. Hence, the pass-through effect increases the marginal benefit of raising w_1 , which puts upward pressure on w_1 .

If v_1 and r_2 are vertically integrated, then—due to the vertical-diversion effect—the price of retail product 2 increases: $p_2^s < p_2^m$. Since retail products are strategic complements, firm f_3 responds by also increasing its retail price, but by less than the retail price of the merged firm: $p_1^s < p_1^m < p_2^m$, where the second inequality follows from equation (2). In addition, due to the vertical-diversion and the pass-through effects, firm f_1 increases manufacturer's price 1: $w_1^s < w_1^{m.140}$ Thus, due to the exercise of market power, if there is a vertical merger, retail prices increase: $\mathbf{p}^s < \mathbf{p}^m$. Social welfare decreases.

In this theory, harm occurs if the vertical merger enables the exercise of market power, just as in a horizontal merger. A necessary condition for harm to occur is that the merged firm transacts with downstream (upstream) rivals: Assumption (A2) ((A3)) holds. This perspective unifies the theories of harm for horizontal and non-horizontal mergers, as we discussed in the Introduction.

5.3.3Vertical Merger and Exit Harm

A firm exits if it has negative profits: If its revenues minus variable costs do not cover its fixed costs. Both the exercise of market power and the efficiencies that are enabled by a vertical merger may cause rivals' exit.

With regard to exit that is caused by efficiencies, as was discussed in Subsection 5.2.2, the EDM causes a decrease in w_1 and a smaller decrease in w_2 , which results in a decrease in p_1 and a smaller decrease in p_2 : $\theta = w_1^e < w_2^e < w_2^s$ and $p_1^e < p_2^e < p_2^s$. These relative retail price variations divert demand from r_2 to r_1 : $D_2(\mathbf{p}^e) < D_2(\mathbf{p}^s)$ and $D_1(\mathbf{p}^s) < D_1(\mathbf{p}^e)$.¹⁴¹ In addition, the decrease in w_2 reduces the profit margin of v_2 . Hence, the profit of firm f_2 decreases.

Consequently, firm f_2 might no longer be able to cover its fixed costs and might exit the industry:

$$\widetilde{\pi}_2^{\upsilon}(\mathbf{w}^e; c_2) - \varphi_{f_2} < 0 \le \widetilde{\pi}_2^{\upsilon}(\mathbf{w}^s; c_2) - \varphi_{f_2}.$$

¹³⁸This vertical-diversion effect flows in the opposite direction of the effect that is caused by an increase in p_2 . ¹³⁹This follows from $b_j^r(\cdot)$ being increasing in $p_{j'}, j' \neq j$, and $\frac{\partial^2 \Pi_{f_1}}{\partial w_1 \partial p_2} > 0$.

¹⁴⁰Firm f_2 may respond by increasing or decreasing w_2 .

 $^{^{141}}$ See Luco and Marshall (2020).

A similar outcome could occur for firm f_4 .

If a vertical merger also generates technological efficiencies, then the decrease of w_1 and p_1 —and thereby of the profits of v_2 and r_4 —is larger. It increases the likelihood that firms f_2 or f_4 may exit after a vertical merger.

With regard to exit that is caused by the exercise of market power, as was discussed in Subsubsection 5.3.2, the exercise of market power causes an increase in w_1 , an increase in p_2 , and a smaller increase in p_1 : $w_1^s < w_1^m$ and $p_1^s < p_1^m < p_2^m$. The increase in retail prices reduces the quantity demanded: $\max\{D_1(\mathbf{p}^m), D_2(\mathbf{p}^m)\}) < D_j(\mathbf{p}^e)$, j=1,2. The increase in w_1 reduces the margin of r_1 . Hence, the profit of firm f_3 decreases. Firm f_3 might no longer be able to cover its fixed costs and might exit the industry. A similar outcome could occur for firm f_2 .¹⁴²

Hence, the efficiencies and the market power that are enabled by a vertical merger reduce the profits of the merged firm's rivals—upstream and downstream—which may cause rivals to exit. If exit occurs, product variety is reduced.¹⁴³ The decrease in product variety might have significant social welfare consequences for consumers.¹⁴⁴ Exit that is caused by the exercise of market power reinforces the direct negative impact of price increases on social welfare.

Exit that is caused by efficiencies reduces, and may reverse, the direct positive effect on social welfare of the price decreases.

Indeed, with exit, *ex-post-merger* prices—even with efficiencies—may not be lower than *ex-ante-merger* prices: On the one hand, *ex-post-merger*, the merged firm has lower marginal costs than do the non-merging firms. On the other hand, with exit, the industry is more concentrated.

5.3.4 Challenges to Prove Market Power and Rivals' Exit Harm

Conditions (A2) and (A3) establish the ability of a merged firm to exercise market power upstream and downstream, respectively. However, to prove harm, it is also necessary to establish the likely substantial magnitude of the exercise of market power.

Horizontal merger review provides many tools and extensive experience from which one can draw to perform this evaluation. By using their legal powers to obtain information, and properly adjusted standard methods of analysis, the Agencies can predict both the exercise of market power and exit, and furthermore, evaluate its impact on social welfare.

These methods of analysis cover a wide range of approaches. At one extreme are soft quantitative methods. They involve using qualitative information or simple data. With information about margins, fixed costs, and expected sale losses, it is possible to evaluate the exercise of market power and identify firms that are at risk of exiting. With data on market shares and diversion ratios, it is possible to obtain rough estimates of the associated price and social welfare

¹⁴²Price w_2 —and thereby the margin of v_2 —may decrease or increase.

¹⁴³When products are differentiated, additional variety increases consumer welfare through better segmentation. In such cases, a reduction in product variety could generate a large decrease in consumer surplus. For a quantification, see, *e.g.*, Donna et al. (2024).

¹⁴⁴See, e.g., Donna, Pereira and Pu (2024).

variations.

At the other extreme are hard quantitative methods. They involve estimating and simulating economic equilibrium models.¹⁴⁵ These methods enable estimating equilibrium prices, profits, and consumer surplus variations, with and without exit. With this information, it is possible to credibly: Quantify price variations; identify firms that are at risk of exiting; and compute the associated social welfare variations.

These are the tools and standards that are used to establish market power harm in horizontal mergers. Hence, this perspective provides a unified approach to evaluating horizontal and vertical mergers.

5.3.5 Challenges to Prove Exclusionary Harm

To determine whether a vertical merger may diminish competition due to foreclosure or raising rivals' costs, it is necessary to establish ability and incentive.

With regard to ability, the issue is whether by refusing to transact with rivals or changing the terms by which it transacts with rivals, the merged firm can cause rivals to lose significant sales, or become less competitive, and eventually to exit the industry. It may depend on whether the rivals have access to equally capable alternative suppliers or clients.¹⁴⁶

With regard to incentive, the issue is whether refusing to transact with rivals—or changing the terms by which it transacts with rivals—is profitable for the merged firm.

These tasks are non-trivial: Identifying alternative suppliers or clients might seem straightforward. However, it depends on the firm's ability and efforts, which are unobservable to outsiders. In turn, evaluating profitability requires enough information to conduct counterfactual exercises. While our theories of harm do not preclude foreclosure or raising rivals' costs effects, they make establishing harm easier. Our analysis above applies also, with the due changes, to mergers of firms that offer complementary products.

Finally, for mergers that involve benefits and harms, computational methods can be useful for a quantification and risk assessment.

¹⁴⁵The industrial organization literature has made substantial progress in the past decades with regard to the development of these methods to quantify, *ex ante*, the likely effect of a merger on prices and social welfare. See, *e.g.*, Nevo (2000), Villas-Boas (2007), Bonnet and Dubois (2010), Crawford et al. (2018), Pereira and Ribeiro (2018), Donna et al. (2022), and Donna et al. (2024).

 $^{^{146}}$ See the discussion of Guideline 5 in Section 2 above.

6 Concluding Remarks

The 2023 Merger Guidelines (MGs) started a much-needed debate about competition and antitrust enforcement policy in the modern economy.¹⁴⁷

Even scholars who disagree about the direction of antitrust enforcement policy might value the outcome of such discussions. Two objectives that were put forward by the drafting team of the MGs are, first, to modernize the Guidelines for current markets and the digital economy, and, second, to address the difficulties that the Agencies faced when challenging mergers in modern markets.

Market definition is a key element that has posed problems for the Agencies in the past decades. The MGs address this difficulty with Guideline 2.¹⁴⁸ Guideline 2 can be used in cases where a precise market definition may be too difficult or where performing a "Hypothetical Monopolist/Monopsonist Test" may not be feasible.¹⁴⁹

The Agencies might nevertheless challenge such mergers based only on Guideline 2 (or combining Guideline 2 with other Guidelines as was discussed in Section 2 above). The Agencies will still have to define the (antitrust) market, perhaps broadly, to comply with the law requirement. However, the challenge's focus will not be based on market definition.¹⁵⁰

The approach is novel and the application is relevant for non-horizontal mergers and vertical markets because the Clayton Act prohibits mergers that may substantially lessen competition or tend to create a monopoly "in any line of commerce and in any section of the country." If the Agencies prove that harm may be substantial in one market, then the merger is illegal.¹⁵¹

We discussed the MGs four-pronged contribution to antitrust enforcement policy in nonhorizontal markets. The MGs remove the horizontal and non-horizontal merger labels. They simplify the narrative with respect to non-horizontal mergers. They expand the breadth and depth of the harm theories in non-horizontal markets. Notably, the MGs expand the reach through which rivals' exit may harm competition in non-horizontal mergers. Finally, they combine economics and law analyses. We provided an introduction to the economics literature that underlyies the MGs and showed that the two are consistent.

Overall, the MGs signal an inflection point in the Agencies' antitrust law enforcement. The review and modernization of the merger guidelines equip the Agencies to apply the recent developments in economic analysis to current industries and the digital economy.

¹⁴⁷There were about 1,600 public comments with regard to the Draft Merger Guidelines that were released on July 19, 2023, according to (accessed January 11, 2024): https://www.regulations.gov/docket/FTC-2023-0043/comments). They include comments from ordinary people, business owners, policymakers, and academics. DOJ and FTC staff participated in a number of panels and conference discussions.

¹⁴⁸"Guideline 2: Mergers Can Violate the Law When They Eliminate Substantial Competition Between Firms." (MGs, p. 2.)

¹⁴⁹These could occur due to a variety of reasons. See Kaplow (2010), Werden (2012), and Kaplow (2013) for discussions.

¹⁵⁰Note that one can still estimate demand, perform a merger simulation, and provide a risk assessment of the merger, even without a precise definition of the relevant antitrust market.

¹⁵¹The recent labor market case—where the Court blocked Penguin Random House's proposed acquisition of Simon & Schuster—provides an example.

It is an open question as to the extent to which the plaintiffs will use these new arguments and whether courts will be convinced. Recent challenges already reflect the Agencies' new approach.

In the forthcoming years, we expect to see more challenges along these lines, a deterrence effect that is signaled by stronger enforcement, and an expansion of the theories of harm in non-horizontal markets.

We have proposed one path forward in this direction, which is consistent with the MGs' economic analysis and current statutory and case law:¹⁵² We discussed the details of the MGs' structural presumptions that involve rivals' exit concerns due to the exercise of market power in non-horizontal mergers. We proposed a structural presumption that recognizes that, in some cases, a vertical merger may reduce competition *even absent any specific anticompetitive intent*.

There has been a heated discussion about the direction of antitrust enforcement policy during the past years. The MGs indicate that merger policy enforcement belongs to the policy domain. They also show that it is paramount to continue using the latest advances from the economics literature to enhance antitrust enforcement policy.

The putative poor merger policy enforcement from the past decades is not the consequence of the shortcomings of economic analysis. It is the consequence of the policy choices by the enforcers.

Denying that mergers have efficiencies, elimination of the double marginalization, or distributional effects that may benefit some parties and harm others would be a baseless negation. Our reading is that the MGs acknowledge these trade-offs. The weights, however, have shifted.

¹⁵²15 U.S.C. §18, Brown Shoe Co. v. U.S., 370 U.S. 294 (1962), and FTC v. Procter & Gamble Co., 386 U.S. 568 (1967).

References

- Alderman, Brianna, Roger D. Blair, and Javier D. Donna. 2024. "The Microsoft Acquisition of Activision: Neither Horizontal nor Vertical." Working Paper, University of Florida.
- Allain, M.L., C. Chambolle, and P. Rey. 2016. "Vertical Integration as a Source of Holdup." The Review of Economic Studies, 83(1): 1–25.
- Allen, Bruce T. 1971. "Vertical integration and market foreclosure: The case of cement and concrete." *The Journal of Law and Economics*, 14(1): 251–274.
- Andrzejewski, Scott. 2023. "Antitrust Complaints Filed Against Ticketmaster and Live Nation in California: An Overview and Analysis." The University of Chicago Business Law Review, Online Article, Available at https://businesslawreview.uchicago.edu/sites/default/files/2023-04/Scott%20Andrzejewski.pdf (Accessed January 11, 2024).
- Antón, Miguel, Florian Ederer, Mireia Giné, and Martin C. Schmalz. 2024. "Innovation: The Bright Side of Common Ownership?" National Bureau of Economic Research Working Paper 32192.
- Antón, Miguel, Florian Ederer, Mireia Giné, and Martin Schmalz. 2023. "Common ownership, competition, and top management incentives." *Journal of Political Economy*, 131(5): 1294–1355.
- Areeda, Phillip E., and Herbert Hovenkamp. 2015-2021. Antitrust Law: An Analysis of Antitrust Principles and Their Application. Fourth and Fifth Editions ed. Originally published in 1978.
- Aryal, G., F. Ciliberto, and B.T. Leyden. 2022. "Coordinated capacity reductions and public communication in the airline industry." *The Review of Economic Studies*, 89(6): 3055– 3084.
- Ashenfelter, O. C., D. S. Hosken, and M. C. Weinberg. 2015. "Efficiencies brewed: pricing and consolidation in the US beer industry." *The RAND Journal of Economics*, 46(2): 328– 361.
- Ashenfelter, Orley C., Daniel S. Hosken, and Matthew C. Weinberg. 2013. "The price effects of a large merger of manufacturers: A case study of Maytag-Whirlpool." *American Economic Journal: Economic Policy*, 5(1): 239–261.
- Asker, John. 2016. "Diagnosing foreclosure due to exclusive dealing." The Journal of Industrial Economics, 64(3): 375–410.
- Asker, John, and Volker Nocke. 2021. "Collusion, mergers, and related antitrust issues." In *Handbook of industrial organization*. Vol. 5, 177–279. Elsevier.

- Athey, Susan, and Fiona Scott Morton. 2022. "Platform Annexation." Antitrust Law Journal, 84(3): 677–703.
- Athey, Susan, Mark Chicu, Malika Krishna, and Ioana Marinescu. 2023. "The Year in Review: Economics at the Antitrust Division, 2022–2023." *Review of Industrial Organization*, 525–544.
- Australian Competition and Consumer Commission. 2022. "Spirit Super and Palisade Investment Partners Consortium decides not to proceed with Port of Geelong acquisition." Australian Competition and Consumer Commission Website, available online (accessed January 11, 2024): https://www.accc.gov.au/media-release/spirit-super-andpalisade-investment-partners-consortium-decides-not-to-proceed-with-portof-geelong-acquisition.
- Autor, David, David Dorn, Lawrence F. Katz, Christina Patterson, and John Van Reenen. 2020. "The fall of the labor share and the rise of superstar firms." The Quarterly Journal of Economics, 135(2): 645–709.
- Azar, José A., Steven T. Berry, and Ioana Marinescu. 2022. "Estimating labor market power." National Bureau of Economic Research, Working Paper No. w30365.
- Azar, José, Ioana Marinescu, and Marshall Steinbaum. 2022. "Labor market concentration." Journal of Human Resources, 57(S): S167–S199.
- Azar, José, Ioana Marinescu, Marshall Steinbaum, and Bledi Taska. 2020. "Concentration in US labor markets: Evidence from online vacancy data." *Labour Economics*, 66: 101886.
- Azar, José, Martin C. Schmalz, and Isabel Tecu. 2018. "Anticompetitive effects of common ownership." The Journal of Finance, 73(4): 1513–1565.
- Backus, Matthew, Christopher Conlon, and Michael Sinkinson. 2019. "The common ownership hypothesis: Theory and evidence." *Economic Studies at Brookings*, 1980–2017.
- Baker, Jonathan B., and Carl Shapiro. 2008. "Reinvigorating Horizontal Merger Enforcement." In How the Chicago School Overshot the Mark: The Effect of Conservative Economic Analysis on U.S. Antitrust. Oxford University Press.
- Bassier, Ihsaan, Arindrajit Dube, and Suresh Naidu. 2022. "Monopsony in movers: The elasticity of labor supply to firm wage policies." *Journal of Human Resources*, 57(S): S50–S86.
- Beck, M., and F. Scott Morton. 2021. "Evaluating the evidence on vertical mergers." *Review* of *Industrial Organization*, 59(2): 273–302.

- Benkard, C Lanier, Ali Yurukoglu, and Anthony Lee Zhang. 2021. "Concentration in product markets." National Bureau of Economic Research.
- Berger, David W., Thomas Hasenzagl, Kyle F. Herkenhoff, Simon Mongey, and Eric A. Posner. 2023. "Merger Guidelines for the Labor Market." National Bureau of Economic Research, Working Paper No. w31147.
- Bernheim, B Douglas, and Michael D Whinston. 1998. "Exclusive dealing." Journal of political Economy, 106(1): 64–103.
- Berry, Steven, Martin Gaynor, and Fiona Scott Morton. 2019. "Do increasing markups matter? Lessons from empirical industrial organization." *Journal of Economic Perspectives*, 33(3): 44–68.
- Bet, Germán. 2021. "Market power in the US airline industry." Available at SSRN 3913695 https://ssrn.com/abstract=3913695 or http://dx.doi.org/10.2139/ssrn.3913695.
- Bet, Germán, Roger D. Blair, and Javier D. Donna. 2022. "The economic rationale of United States v. Google." *The Antitrust Bulletin*, 67(1): 23–39.
- Bhattacharya, V., G. Illanes, and D. Stillerman. 2023. "Merger Effects and Antitrust Enforcement: Evidence from US Retail." National Bureau of Economic Research, No. w31123.
- Bonnet, Céline, and Pierre Dubois. 2010. "Inference on vertical contracts between manufacturers and retailers allowing for nonlinear pricing and resale price maintenance." The RAND Journal of Economics, 41(1): 139–164.
- Bork, Robert H. 1978. The Antitrust Paradox: A Policy at War with Itself. Bork Publishing. Published in 2021. Originally published in 1978.
- Brown, Z.Y., and A. MacKay. 2023. "Competition in pricing algorithms." American Economic Journal: Microeconomics, 15(2): 109–156.
- Bulow, Jeremy I, John D. Geanakoplos, and Paul D. Klemperer. 1985. "Multimarket oligopoly: Strategic substitutes and complements." *Journal of Political economy*, 93(3): 488–511.
- Cabral, Luís. 2021. "Merger policy in digital industries." *Information Economics and Policy*, 54: 100866. Antitrust in the Digital Economy.
- Caradonna, Peter, Nathan Miller, and Gloria Sheu. 2023. "Mergers, Entry, and Consumer Welfare." *Georgetown McDonough School of Business Research Paper No. 3537135*.
- Chipty, Tasneem. 2001. "Vertical integration, market foreclosure, and consumer welfare in the cable television Industry." *American Economic Review*.

- Choi, J. Pil, and B.C. Kim. 2010. "Net neutrality and investment incentives." *RAND Journal* of *Economics*, 41(3): 446–471.
- Cho, Inkoo, and Noah Williams. 2024. "Collusive Outcomes Without Collusion." arXiv. DOI: 10.48550/arXiv.2403.07177.
- Ciliberto, Federico, Charles Murry, and Elie Tamer. 2021. "Market structure and competition in airline markets." *Journal of Political Economy*, 129(11): 2995–3038.
- Collins, Dale. 2016. "Antitrust Law, Unit 9: Introduction to Mergers." NYU School of Law, Available at https://appliedantitrust.com/09_merger_guidelines/unit9_materials2016.pdf (Accessed January 11, 2024).
- Cooper, Zack, Stuart V. Craig, Martin Gaynor, and John Van Reenen. 2019. "The price ain't right? Hospital prices and health spending on the privately insured." *The Quarterly Journal of Economics*, 134(1): 51–107.
- Corbett, Charles J., Deming Zhou, and Christopher S. Tang. 2004. "Designing Supply Contracts: Contract Type and Information Asymmetry." *Management Science*, 50(4): 550–559.
- Cournot, Antoine Augustin. 1838. Recherches sur les principes mathématiques de la théorie des richesses. L. Hachette.
- Crawford, Gregory S., Robin S. Lee, Michael D. Whinston, and Ali Yurukoglu. 2018. "The welfare effects of vertical integration in multichannel television markets." *Econometrica*, 86(3): 891–954.
- Cunningham, C., F. Ederer, and S. Ma. 2021. "Killer acquisitions." Journal of Political Economy, 129(3): 649–702.
- Debreu, Gerard. 1970. "Economies with a finite set of equilibria." Econometrica, 387–392.
- **De Corniere, Alexandre, and Greg Taylor.** 2019. "A model of biased intermediation." *The RAND Journal of Economics*, 50(4): 854–882.
- **De Loecker, Jan, Jan Eeckhout, and Gabriel Unger.** 2020. "The rise of market power and the macroeconomic implications." *Quarterly Journal of Economics*, 135(2): 561–644.
- **Donna, Javier D.** 2021. "Measuring long-run gasoline price elasticities in urban travel demand." *The RAND Journal of Economics*, 52(4): 945–994.
- Donna, Javier D., and Anita N. Walsh. 2023. "(Lack of) Competition, Coordination, and Information Sharing in the Pork Industry: United States, 2009–2020." *The Antitrust Bulletin*, 68(1): 117–136.

- **Donna, Javier D., and José-Antonio Espín-Sánchez.** 2024. "The illiquidity of water markets." *Review of Economic Studies*, (forthcoming). *Available at SSRN 2667654*.
- **Donna, Javier D., and Pedro Pereira.** 2023*a*. "Rivals' Exit and Vertical Merger Evaluation." Journal of Competition Law & Economics, DOI: 10.1093/joclec/nhad002, 19(2): 220–249.
- Donna, Javier D., and Pedro Pereira. 2023b. "Rivals' Exit Should Be Incorporated into the Guidelines for Vertical Merger Evaluation." ProMarket, Stigler Center, University of Chicago, May 2023, available online (accessed January 11, 2024): https://www.promarket.org/2023/05/22/rivals-exit-should-be-incorporatedinto-the-guidelines-for-vertical-merger-evaluation/.
- Donna, Javier D., and Pedro Pereira. 2023c. "An Anticompetitive Presumption for Exit-Inducing Vertical Arrangements." *Draft Merger Guidelines for Public Comment by the Federal Trade Commission*, Docket ID: FTC-2023-0043, Tracking Number: lm7-ttk2-thh7, Available at SSRN 4709489, http://dx.doi.org/10.2139/ssrn.4709489.
- **Donna, Javier D., Pedro Pereira, and Yun Pu.** 2024. "Exit-inducing Vertical Mergers." Working Paper, University of Florida.
- Donna, Javier D., Pedro Pereira, Tiago Pires, and André Trindade. 2022. "Measuring the welfare of intermediaries." *Management Science*, 68(11): 8083–8115.
- Donna, Javier D., Pedro Pereira, Yun Pu, Andre Trindade, and Renan C Yoshida. 2024. "Direct Sales and Bargaining." *RAND Journal of Economics*, (forthcoming), *Available at SSRN 3779962*.
- **Dubé**, Jean-Pierre, Günter J. Hitsch, and Peter E. Rossi. 2009. "Do switching costs make markets less competitive?" *Journal of Marketing Research*, 46(4): 435–445.
- **Dubé, Jean-Pierre, Günter J Hitsch, and Peter E Rossi.** 2010. "State dependence and alternative explanations for consumer inertia." *The RAND Journal of Economics*, 41(3): 417–445.
- **Eeckhout, Jan.** 2022. The Profit Paradox: How Thriving Firms Threaten the Future of Work. Princeton University Press.
- Elhauge, Einer. 2020. "How horizontal shareholding harms our economy-and why antitrust law can fix it." *Harvard Business Law Review*, 10: 207.
- Fan, Y., and C. Yang. 2020. "Competition, product proliferation, and welfare: A study of the US smartphone market." American Economic Journal: Microeconomics, 12(2): 99–134.
- Fan, Y., and C. Yang. 2022. "Estimating discrete games with many firms and many decisions: An application to merger and product variety." National Bureau of Economic Research, No. w30146.
- Farrell, Joseph, and Carl Shapiro. 2000. "Scale economies and synergies in horizontal merger analysis." *Antitrust Law Journal*, 68: 685.
- Farrell, Joseph, and Paul Klemperer. 2007. "Coordination and lock-in: Competition with switching costs and network effects." Handbook of industrial organization, 3: 1967–2072. Federal Trade Commission and Plaintiff States v. Amazon.com, Inc.
- Federal Trade Commission and Plaintiff States v. Amazon.com, Inc. 2023. "District Court of Western District of Washington." Case 2:23-cv-01495-JHC Document 114 Filed 11/02/23.
- Federal Trade Commission Complaint. 2022. "In the Matter of Microsoft Corp. and Activision Blizzard Inc., United States of America Before the Federal Trade Commission." Docket No.9412, file number 2210077.
 Ended Ender Commission Ended

Federal Trade Commission v. Facebook

- Federal Trade Commission v. Facebook. 2021. "First Amended Complaint For Injunctive and Other Equitable Relief, United States District Court for the District Of Columbia." Case No.: 1:20-cv-03590-JEB, Filed 08/19/21. Available at: https://www.ftc.gov/ system/files/documents/cases/ecf_75-1_ftc_v_facebook_public_redacted_fac.pdf (Accessed January 11, 2024).
- Federico, G., G. Langus, and T. Valletti. 2018. "Reprint of: Horizontal mergers and product innovation." *International Journal of Industrial Organization*, 61: 590–612.
- Fudenberg, Drew, and Jean Tirole. 1991. Game theory. MIT press.
- **Fumagalli, Chiara, and Massimo Motta.** 2006. "Exclusive Dealing and Entry, when Buyers Compete." *American Economic Review*, 96(3): 785–795.
- Gal-Or, Esther. 1991. "Vertical Restraints with Incomplete Information." The Journal of Industrial Economics, 39(5): 503–516.
- Grieco, P., J. Pinkse, and M. Slade. 2018. "Brewed in North America: Mergers, marginal costs, and efficiency." *International Journal of Industrial Organization*, 59: 24–65.
- Gutiérrez, Germán, and Thomas Philippon. 2017. "Declining Competition and Investment in the US." NBER Working paper No. w23583, National Bureau of Economic Research.
- Hart, Oliver D., and Jean Tirole. 1990. "Vertical integration and market foreclosure." Brookings Papers on Economic Activity, 205–276.
- Hastings, J.S. 2004. "Vertical relationships and competition in retail gasoline markets: Empirical evidence from contract changes in Southern California." *American Economic Review*, 94(1): 317–328.

- Hemphill, C. Scott, and Nancy L. Rose. 2017. "Mergers that harm sellers." Yale Law Journal, 127: 2078.
- Hortaçsu, Ali, and Chad Syverson. 2007. "Cementing relationships: Vertical integration, foreclosure, productivity, and prices." *Journal of Political Economy*, 115(2): 250–301.
- Hortaçsu, Ali, Seyed Ali Madanizadeh, and Steven L Puller. 2017. "Power to choose? An analysis of consumer inertia in the residential electricity market." *American Economic Journal: Economic Policy*, 9(4): 192–226.
- Houde, J.-F. 2012. "Spatial differentiation and vertical mergers in retail markets for gasoline." American Economic Review, 102(5): 2147–2182.
- Hovenkamp, Erik. 2018. "Platform Antitrust." J. Corp. L., 44: 713.
- Hovenkamp, Erik. 2023. "The Competitive Effects of Search Engine Defaults." USC CLASS Research Paper No. 24-10, Available at SSRN 4647211.
- Igami, M., and T. Sugaya. 2022. "Measuring the incentive to collude: the vitamin cartels, 1990–99." The Review of Economic Studies, 89(3): 1460–1494.
- Jullien, B., A. Pavan, and M. Rysman. 2021. "Two-sided markets, pricing, and network effects." In *Handbook of Industrial Organization*. Vol. 4, 485–592. Elsevier.
- Kamepalli, Sai Krishna, Raghuram Rajan, and Luigi Zingales. 2020. "Kill zone." National Bureau of Economic Research.
- Kaplow, L., and C. Shapiro. 2007. "Antitrust." Harvard John M. Olin Center For Law, Economics, And Business, Discussion Paper No. 575, ISSN 1045-6333, available at (accessed January 11, 2024): http://www.law.harvard.edu/programs/olin_center/papers/pdf/ Kaplow_Shapiro_575.pdf.
- Kaplow, Louis. 2010. "Why (ever) define markets." Harvard Law Review, 124: 437.
- Kaplow, Louis. 2013. "Market definition: Impossible and counterproductive." Antitrust Law Journal, 79(361): 361.
- Klemperer, Paul. 1987a. "The competitiveness of markets with switching costs." The RAND Journal of Economics, 138–150.
- Klemperer, Paul. 1987b. "Markets with consumer switching costs." The Quarterly Journal of Economics, 102(2): 375–394.
- Kwoka, John. 2015. Mergers, merger control, and remedies: A retrospective analysis of US policy. MIT Press.

- Kwoka, John. 2016. "The structural presumption and the safe harbor in merger review: False positives or unwarranted concerns." *Antitrust Law Journal*, 81: 837.
- Kwoka, John, and Evgenia Shumilkina. 2010. "The price effect of eliminating potential competition: Evidence from an airline merger." The journal of industrial economics, 58(4): 767–793.
- Kwoka, John, and Margaret Slade. 2019. "Second thoughts on double marginalization." *Antitrust*, 34: 51.
- Lancieri, Filippo, Eric A Posner, and Luigi Zingales. 2022. "The Political Economy of the Decline of Antitrust Enforcement in the United States." National Bureau of Economic Research.
- Lee, Robin S., Michael D. Whinston, and Ali Yurukoglu. 2021. "Structural empirical analysis of contracting in vertical markets." In *Handbook of Industrial Organization*. Vol. 4, 673–742. Elsevier.
- Letina, Igor, Armin Schmutzler, and Regina Seibel. 2021. "Killer Acquisitions and Beyond: Policy Effects on Innovation Strategies." University of Zurich, Department of Economics, Working Paper No. 358, Revised version, Available at http://dx.doi.org/10.2139/ ssrn.3673150.
- Li, S., J. Mazur, J. Roberts, and A. Sweeting. 2015. "Airline mergers and the potential entry defense." Discussion paper, University of Maryland, available at: https://econweb. umd. edu/~sweeting/SWEETING_PE2016.pdf.
- Luco, Fernando, and Guillermo Marshall. 2020. "The competitive impact of vertical integration by multiproduct firms." *American Economic Review*, 110(7): 2041–2064.
- Marinescu, Ioana, and Herbert Hovenkamp. 2019. "Anticompetitive mergers in labor markets." *Indiana Law Journal*, 94: 1031.
- Marvel, Howard P. 1982. "Exclusive dealing." Journal of Law and Economics, 25(1): 1–25.
- Miller, N.H., and M.C. Weinberg. 2017. "Understanding the price effects of the MillerCoors joint venture." *Econometrica*, 85(6): 1763–1791.
- Miller, N.H., G. Sheu, and M.C. Weinberg. 2021. "Oligopolistic price leadership and mergers: The United States beer industry." *American Economic Review*, 111(10): 3123–3159.
- Moresi, Serge, and Steven C. Salop. 2021. "When vertical is horizontal: How vertical mergers lead to increases in "effective concentration"." *Review of Industrial Organization*, 59: 177–204.

- Moss, Diana L. 2020. "What Does Expanding Horizontal Control Mean for Antitrust Enforcement? A Look at Mergers, Partial Ownership, and Joint Ventures." A Look at Mergers, Partial Ownership, and Joint Ventures (November 4, 2020). Available at SSRN: 3860363.
- Motta, Massimo. 2004. Competition Policy: Theory and Practice. Cambridge University Press.
- Naidu, Suresh. 2010. "Recruitment restrictions and labor markets: Evidence from the postbellum US South." *Journal of Labor Economics*, 28(2): 413–445.
- Naidu, Suresh, and Eric A Posner. 2022. "Labor Monopsony and the Limits of the Law." *Journal of Human Resources*, 57(S): S284–S323.
- Naidu, Suresh, Eric A Posner, and Glen Weyl. 2018. "Antitrust remedies for labor market power." *Harvard Law Review*, 132(2): 536–601.
- Naidu, Suresh, Yaw Nyarko, and Shing-Yi Wang. 2016. "Monopsony power in migrant labor markets: evidence from the United Arab Emirates." *Journal of Political Economy*, 124(6): 1735–1792.
- Naidu, Suresh, Yaw Nyarko, and Shing-Yi Wang. 2023. "The Benefits and Costs of Guest Worker Programs: Experimental Evidence from the India-UAE Migration Corridor." National Bureau of Economic Research.
- **Nevo, Aviv.** 2000. "Mergers with Differentiated Products: The Case of the Ready-to-Eat Cereal Industry." *The RAND Journal of Economics*, 31(3): 395–421.
- Nocke, Volker, and Lucy White. 2007. "Do vertical mergers facilitate upstream collusion?" *American Economic Review*, 97(4): 1321–1339.
- Nocke, Volker, and Michael D Whinston. 2010. "Dynamic merger review." Journal of Political Economy, 118(6): 1200–1251.
- Nocke, Volker, and Michael D. Whinston. 2022. "Concentration Thresholds for Horizontal Mergers." *American Economic Review*, 112(6): 1915–48.
- Nocke, Volker, and Patrick Rey. 2018. "Exclusive dealing and vertical integration in interlocking relationships." *Journal of Economic Theory*, 177: 183–221.
- **OECD.** 2017. "Common Ownership by Institutional Investors and its Impact on Competition, Background Note by the Secretariat." Organisation for Economic Co-operation and Development Website, available online (accessed January 11, 2024): https://one.oecd. org/document/DAF/COMP(2017)10/en/pdf.
- **Ordover, Janusz A., Garth Saloner, and Steven C. Salop.** 1990. "Equilibrium vertical foreclosure." *American Economic Review*, 80(1): 127–142.

- Park, Alex Haerang, and Kyoungwon Seo. 2019. "Common ownership and product market competition: evidence from the us airline industry." *Korean Journal of Financial Studies*, 48(5): 617–640.
- **Pereira, Pedro, and Tiago Ribeiro.** 2018. "Evaluating partial divestitures when vertical relations are important." *Review of Industrial Organization*, 53(2): 321–345.
- Peterson, Christopher L, and Marshall Steinbaum. 2023. "Coercive rideshare practices: At the intersection of antitrust and consumer protection law in the gig economy." *University* of Chicago Law Review, 90: 623.
- Prager, Elena, and Matt Schmitt. 2021. "Employer consolidation and wages: Evidence from hospitals." American Economic Review, 111(2): 397–427.
- ProMarket Writers. 2024. "The Trends and Cases That Will Define US Antitrust in 2024." Stigler Center, University of Chicago, January 2024. Available at: https://www.promarket.org/2024/01/05/the-trends-and-cases-that-willdefine-us-antitrust-in-2024/ (Accessed January 11, 2024).
- Rasmusen, Eric B., J. Mark Ramseyer, and John S. Wiley. 1991. "Naked Exclusion." The American Economic Review, 81(5): 1137–1145.
- Reimers, Imke, and Joel Waldfogel. 2023. "A Framework for Detection, Measurement, and Welfare Analysis of Platform Bias." National Bureau of Economic Research Working Paper 31766.
- Rey, Patrick, and Jean Tirole. 2007. "A Primer on Foreclosure." In *Handbook of Industrial Organization*. Vol. 3, , ed. Mark Armstrong and Robert H. Porter, 2145–2220.
- Riordan, Michael H, and Steven C. Salop. 1995. "Evaluating Vertical Mergers: A Post-Chicago Approach." Antitrust Law Journal, 63(2): 513–568.
- **Rogerson, William P.** 2020. "Modelling and predicting the competitive effects of vertical mergers: The bargaining leverage over rivals effect." *Canadian Journal of Economics/Revue canadienne d'économique*, 53(2): 407–436.
- Rogerson, William P. 2021. "The Upstream Pass-Through Rate, Bargaining Power and the Magnitude of the Raising Rivals' Costs (RRC) Effect." *Review of Industrial Organization*, 59(2): 205–227.
- **Rysman, M.** 2009. "The economics of two-sided markets." *Journal of economic perspectives*, 23(3): 125–143.
- Salinger, Michael A. 1988. "Vertical mergers and market foreclosure." *The Quarterly Journal* of *Economics*, 103(2): 345–356.

- Salop, Steven C. 1981. "Strategy, predation, and antitrust analysis." Federal Trade Commission, Bureau of Economics, Bureau of Competition, Available at https://books.google.com/books?id=LyW7AAAAIAAJ (Accessed March 2, 2022).
- Salop, Steven C. 2015. "The evolution and vitality of merger presumptions: A decision-theoretic approach." Antitrust Law Journal, 80(2): 269–306.
- Salop, Steven C. 2018. "Invigorating Vertical Merger Enforcement." Available at http://www.jstor.org/stable/45222589.
- Salop, Steven C. 2020. "The 2020 Vertical Merger Guidelines: A Suggested Revision." Available at: https://www.ftc.gov/system/files/attachments/798-draft-verticalmerger-guidelines/salop_suggested_vertical_merger_guidelines.pdf (Accessed January 11, 2024).
- Salop, Steven С. 2023. "An Excessive Evidentiary Burden Sunk the FTC's Meta/Within Merger." ProMarket, Case Against the Stigler Center, University of Chicago, February 2023, available online (accessed January 11, 2024):https://www.promarket.org/2023/02/22/an-excessive-evidentiary-burden-sunkthe-ftcs-case-against-the-meta-within-merger/.
- Salop, Steven C., and David T. Scheffman. 1983. "Raising rivals' costs." The American economic review, 73(2): 267–271.
- Scheffman, David, and Richard S. Higgins. 2014. "Raising Rivals' Costs." In Oxford Handbook on International Antitrust Economics., ed. Roger D. Blair and D. Daniel Sokol, 62. Oxford University Press.
- Schmalz, Martin C. 2018. "Common-ownership concentration and corporate conduct." Annual Review of Financial Economics, 10: 413–448.
- Schmalz, Martin C. 2021. "Recent studies on common ownership, firm behavior, and market outcomes." *The Antitrust Bulletin*, 66(1): 12–38.
- Scott Morton, Fiona M., and D. C. Dinielli. 2022. "Roadmap for an Antitrust Case Against Facebook." *Stanford Journal of Law, Business & Finance*, 27(2): 267–316.
- Segal, Ilya R., and Michael D. Whinston. 2000. "Naked Exclusion: Comment." American Economic Review, 90(1): 296–309.
- Shapiro, Carl. 2018. "Antitrust in a Time of Populism." International Journal of Industrial Organization, 61: 714–748.
- Shapiro, Carl. 2019. "Protecting Competition in the American Economy: Merger Control, Tech Titans, Labor Markets." Journal of Economic Perspectives, 33(3): 69–93.

- Shapiro, Carl. 2021. "Vertical Mergers and Input Foreclosure Lessons from the AT&T/Time Warner Case." *Review of Industrial Organization*, 59(2): 303–341.
- Shcherbakov, Oleksandr. 2016. "Measuring consumer switching costs in the television industry." The RAND Journal of Economics, 47(2): 366–393.
- **Shum, Matthew.** 2004. "Does advertising overcome brand loyalty? Evidence from the breakfast-cereals market." *Journal of Economics & Management Strategy*, 13(2): 241–272.
- Singer, Hal. 2023. "The New Merger Guidelines Offer an Important Course Correction on Merger Enforcement." The Sling, Economic Analysis and Competition Policy Research, February 2023, available online (accessed January 11, 2024): https://www.thesling.org/the-new-merger-guidelines-offer-an-important-course-correction-on-merger-enforcement.
- Sixty-Third Congress. 1914. "Clayton Act." Sees. II., Ch. 323, Available at https: //govtrackus.s3.amazonaws.com/legislink/pdf/stat/38/STATUTE-38-Pg730.pdf (Accessed January 11, 2024).
- **Spengler, Joseph J.** 1950. "Vertical integration and antitrust policy." *Journal of political* economy, 58(4): 347–352.
- Sullivan, Sean P. 2016. "What Structural Presumption: Reuniting Evidence and Economics on the Role of Market Concentration in Horizontal Merger Analysis." *Journal of Corporation Law*, 42: 403.
- Tamayo, Jorge Andrés, and Guofu Tan. 2021. Competitive two-part tariffs. Harvard Business School.
- **Topkis, Donald M.** 1998. Supermodularity and complementarity. Princeton university press. United States and Plaintiff States v. JetBlue Airways Corporation and Spirit Airlines, Inc.
- United States and Plaintiff States v. JetBlue Airways Corporation and Spirit Airlines, Inc. 2024. "District Court of Massachusetts." Case 1:23-cv-10511-WGY, Document 461, Filed 01/16/24. Available at: https://oag.ca.gov/system/files/attachments/ press-docs/JetBlue%20Spirit.pdf (Accessed January 18, 2024). United States of America et al. v. Google LLC
- United States of America et al. v. Google LLC. 2020. "Case Number: 1:20-cv-03010."
 District Of Columbia District Court, Filed: 10/20/2020.
 United States v. Bertelsmann Se & Co. Kgaa, Penguin Random House, Llc, Viacomcbs, Inc., and Simon & Schuster, Inc.
- United States v. Bertelsmann Se & Co. Kgaa, Penguin Random House, Llc,

Viacomcbs, Inc., and Simon & Schuster, Inc. 2022. "District Court for the District Of Columbia." Case 1:21-cv-02886-FYP Document 194 Filed 11/07/22. Available at: https://www.justice.gov/atr/case-document/file/1549941/download (Accessed January 11, 2024).

United States v. Sabre Corporation, Sabre Glbl Inc., Farelogix, Inc., and Sandler Capital Partners V, L.P.

- United States v. Sabre Corporation, Sabre Glbl Inc., Farelogix, Inc., and Sandler Capital Partners V, L.P. 2019. "District Court for the District Of Delaware." Case 1:99-mc-09999, Document 1310, Filed 08/20/19. Available at: https://www.justice.gov/ media/1022536/dl (Accessed January 11, 2024).
- U.S. Department of Justice and the Federal Trade Commission. 2010. "Horizontal Merger Guidelines." Available online (accessed on February 8, 2022): https://www.justice.gov/atr/public/guidelines/hmg-2010.pdf.
- U.S. Department of Justice and the Federal Trade Commission. 2020. "Vertical Merger Guidelines." Available online (accessed on February 8, 2022): https: //www.ftc.gov/system/files/documents/reports/us-department-justice-federaltrade-commission-vertical-merger-guidelines/vertical_merger_guidelines_6-30-20.pdf.
- U.S. Department of Justice and the Federal Trade Commission. 2023. "Merger Guidelines." Issued: December 18, 2023, available online (accessed December 18, 2023): https: //www.ftc.gov/system/files/ftc_gov/pdf/2023_merger_guidelines_final_12.18. 2023.pdf.

U.S. Department of Justice v. Agri Stats Inc.

- U.S. Department of Justice v. Agri Stats Inc. 2023. "United States District Court For The District Of Minnesota." Civil Action. CASE 0:23-cv-03009 Doc. 1. Filed 09/28/23.
- Villas-Boas, Sofia Berto. 2007. "Vertical relationships between manufacturers and retailers: Inference with limited data." *The Review of Economic Studies*, 74(2): 625–652.
- Waldfogel, Joel. 2024. "Amazon Self-preferencing in the Shadow of the Digital Markets Act." National Bureau of Economic Research Working Paper 32299.
- Wang, Jian-Cai, Amy Hing-Ling Lau, and Hon-Shiang Lau. 2012. "Practical and effective contracts for the dominant retailer of a newsvendor product with price-sensitive demand." *International Journal of Production Economics*, 138(1): 46–54.
- Weise, Karen, C. Metz, N. Grant, and M. Isaac. 2023. "Inside the A.I. Arms Race That Changed Silicon Valley Forever." *The New York Times*. Available at: https://www.nytimes.

com/2023/12/05/technology/ai-chatgpt-google-meta.html (Accessed on February 8, 2022).

- Werden, Gregory J. 2012. "Why (Ever) Define Markets: An Answer to Professor Kaplow." Antitrust LJ, 78: 729.
- Williamson, Oliver E. 1968. "Economies as an antitrust defense: correction and reply." *The American Economic Review*, 58(5): 1372–1376.
- Wollmann, Thomas G. 2019. "Stealth consolidation: Evidence from an amendment to the Hart-Scott-Rodino Act." American Economic Review: Insights, 1(1): 77–94.
- Wollmann, Thomas G. 2020. "How to get away with merger: Stealth consolidation and its effects on us healthcare." *National Bureau of Economic Research*, Working Paper No. w27274.



Figure 1: Separated Firms.

Figure 2: Vertical Merger and Elimination of Double Marginalization.





Figure 3: Vertical Merger and Market Power.

Appendix

A Exit Concerns in the 2023 Merger Guidelines

The 2023 Merger Guidelines incorporate several anticompetitive presumptions and concerns with regard mergers that may cause rivals to exit. Below is a summary:

Guideline 1 (page 5). "In highly concentrated markets, a merger that eliminates a significant competitor creates significant risk that the merger may substantially lessen competition or tend to create a monopoly." (Emphasis added.)

Guideline 1 (page 6, footnote 11). "Similar concerns arise if the merger threatens to *cause* the exit of a current market participant, such as a leveraged buyout that puts the target firm at significant risk of failure." (Emphasis added.)

Guideline 2 (page 7). "Prior Merger, Entry, and *Exit Events*. The Agencies may look to historical events to assess the presence and substantiality of direct competition between the merging firms. For example, the Agencies may examine the competitive impact of recent relevant mergers, entry, expansion, or *exit events*." (Emphasis added.)

Guideline 3 (page 9). "Elimination of a Maverick. [...] A merger that eliminates a maverick or significantly changes its incentives increases the susceptibility to coordination." (Emphasis added.)

Guideline 3 (pages 9-10). "Aligned Incentives. *Removing a firm* that has different incentives from most other firms in a market can increase the risk of coordination. For example, a firm with a small market share may have less incentive to coordinate because it has more to gain from winning new business than other firms." (Emphasis added.)

Guideline 4 (page 12). "Likely Influence on Existing Rivals. Direct evidence that the firm's presence or behavior has affected or is affecting current market participants' strategic decisions is not necessary but can establish a showing of a likely influence. Even without such direct evidence, circumstantial evidence that the firm's presence or behavior had an effect on the competitive reactions of firms in the market may also show likely influence." (Emphasis added.) Whereas the excerpt does not focus on exit events, it encompasses them. For example, see the argument in FTC v. Procter & Gamble, 386 U.S. 568, 581 (1967): "Third, the number of potential entrants was not so large that the elimination of one would be insignificant." (Emphasis added.)

Guideline 5 (page 14). "Rivals or potential rivals may be excluded from the relevant market if limiting their access to the related product *could lead them to exit the market*." (Emphasis added.)

Guideline 6 (page 20). "*Eliminating a Nascent Competitive Threat.* A merger may involve a dominant firm acquiring a nascent competitive threat—namely, a firm that could grow into a significant rival, facilitate other rivals' growth, or otherwise lead to a reduction in its power." (Emphasis added.)

Guideline 9 (pages 25-26, footnote 48). "However, few participants will leave if, for example, the switching costs are relatively high or if the advantaged product is a small component of the overall set of services those participants access on the platform. Moreover, in the long run few participants will leave if scale economies, network effects, or entry barriers enable the advantaged product to eventually gain market power of its own, with rivals of the advantaged product exiting or becoming less attractive. After these dynamics play out, the platform operator could advantage its own products without losing as many participants, as there would be fewer alternative products available through other channels." (Emphasis added.)

Rebuttal evidence (page 31). "The Agencies evaluate *evidence of a failing firm* consistent with this prevailing law." (Emphasis added.) Footnote 64: "The Agencies do not normally credit claims that the assets of a division would *exit the relevant market* in the near future unless: (1) applying cost allocation rules that reflect true economic costs, the division has a persistently negative cash flow on an operating basis, and such negative cash flow is not economically justified for the firm by benefits such as added sales in complementary markets or enhanced customer goodwill; and (2) the owner of the failing division has made unsuccessful good-faith efforts to elicit reasonable alternative offers that would keep its assets in the relevant market and pose a less severe danger to competition than does the proposed acquisition. Because firms can allocate costs, revenues, and intra-company transactions among their subsidiaries and divisions, the Agencies require evidence that is not solely based on management plans that could have been prepared for the purpose of demonstrating negative cash flow or the prospect of *exit from the relevant market*." (Emphasis added.)

Rebuttal evidence (page 31). "Entry and repositioning. [...] Moreover, the entry must be durable: an entrant that does not plan to sustain its investment or *that may exit the market* would not ensure long-term preservation of competition." (Emphasis added.)

Analytical, Economic, and Evidentiary Tools (Subsection 4.1, page 34). "Evidence that the merging parties intend or expect the merger to lessen competition, such as plans to coordinate with other firms, raise prices, reduce output or capacity, reduce product quality or variety, lower wages, cut benefits, *exit a market*, cancel plans to enter a market without a

merger, withdraw products or delay their introduction, or curtail research and development efforts after the merger, can be highly informative in evaluating the effects of a merger on competition. The Agencies give little weight, however, to the lack of such evidence or the expressed contrary intent of the merging parties." (Emphasis added.)

Analytical, Economic, and Evidentiary Tools (Subsection 4.2, page 36). "Prior Merger, Entry, and *Exit Events*. The Agencies may look to historical events to assess the presence and substantiality of direct competition between the merging firms. For example, the Agencies may examine the impact of recent relevant mergers, entry, expansion, or exit events on the merging parties or their competitive behavior." (Emphasis added.)

Analytical, Economic, and Evidentiary Tools (Subsection 4.3, page 42). "Benchmark for the SSNIPT. The HMT asks whether the hypothetical monopolist likely would worsen terms relative to those that likely would prevail absent the proposed merger. In some cases, the Agencies will use as a benchmark different outcomes than those prevailing prior to the merger. For example, if outcomes are likely to change absent the merger, *e.g.*, because of innovation, entry, *exit*, or exogenous trends, the Agencies may use anticipated future outcomes as the benchmark. Or, if suppliers in the market are coordinating prior to the merger, the Agencies may use a benchmark that reflects conditions that would arise if coordination were to break down. When evaluating whether a merging firm is dominant (Guideline 6), the Agencies may use terms that likely would prevail in a more competitive market as a benchmark." (Emphasis added.)

B Horizontal Concentration and Market Power

Here we present a version of the model that is useful for understanding the relationship between a horizontal merger and market power.

Consider the model of Section 4: Let firms f_1 and f_2 own v_1 and v_2 , respectively; firm f_3 owns $\{r_1, r_2\}$; and firm f_4 is inactive.¹⁵³ The merged firm— f_3 —owns two substitute retail products. Denote equilibrium values by superscript "h". For $f = f_3$ and j = 1, 2, the equilibrium conditions for p_j is:

$$\frac{\partial \Pi_{f_1}}{\partial p_j} = \underbrace{D_j}_{\text{margin effect}>0} + \underbrace{(p_j^h - w_j)\frac{\partial D_j}{\partial p_j}}_{\text{volume-of-sales effect} < 0} + \underbrace{(p_{j'}^h - w_{j'})\frac{\partial D_{j'}}{\partial p_j}}_{\text{horizontal-diversion effect} > 0} = 0$$

The equilibrium condition for p_j , j = 1, 2, includes the margin effect, the volume-of-sales effect, and an additional effect: the *horizontal-diversion effect*. An increase in p_j diverts some sales from r_j to $r_{j'}$, and thereby increases the latter firm's profit. Since the firm f_3 owns both products, this profit increase is appropriated by the firm. The horizontal-diversion effect decreases the marginal cost of raising p_j , which puts upward pressure on retail prices.

For $f = f_1, f_2$ and j = 1, 2, the equilibrium conditions for w_j are as in equation (5). As a consequence, retail prices increase: $\mathbf{p}^e < \mathbf{p}^h$. Social welfare decreases. A similar process occurs if the same firm owns $\{v_1, v_2\}$ —the complements of the two substitute retail products.

¹⁵³Hence, $\Omega^{\upsilon\upsilon} = \mathbb{I}_2$, $\Omega^{rr} = \mathbb{U}_2$, and $\Omega^{\tau\tau'} = \mathbb{O}_2$ if $\tau \neq \tau'$; $\tau, \tau' = \upsilon, r$.